IET Renewable Power Generation

Special Issue Call for Papers

Be Seen. Be Cited. Submit your work to a new IET special issue

Connect with researchers and experts in your field and share knowledge.

Be part of the latest research trends, faster.

# **Read more**





#### REVIEW

### **IET Renewable Power Generation**

Check for updates

The Institution of Engineering and Technology WILEY

# A review of wave energy technology from a research and commercial perspective

# Bingyong Guo<sup>1,2</sup> D | John V. Ringwood<sup>2</sup>

Revised: 15 August 2021

<sup>1</sup> School of Marine Science and Technology, Northwestern Polytechnical University, Xi'an, Shaanxi, China

<sup>2</sup> Centre for Ocean Energy Research, Maynooth University, Maynooth, Co. Kildare, Ireland

#### Correspondence

Bingyong Guo, School of Marine Science and Technology, Northwestern Polytechnical University, Xi'an, Shaanxi 710072, China, Email: bingyongguo@outlook.com

Funding information

H2020 Marie Skłodowska-Curie Actions, Grant/Award Number: 841388; Science Foundation Ireland, Grant/Award Numbers: SFI/13/IA/1886, 12/RC/2302-P2

#### **INTRODUCTION** 1

'Carbon neutrality by 2050' is the world's most urgent mission, and António Guterres, the United Nations Secretary General, stressed on 11 December 2020 [1] that "By next month, countries representing more than 65 per cent of harmful greenhouse gasses and more than 70 per cent of the world economy will have committed to achieve net zero emissions by the middle of the century." To date, more than 110 countries have pledged to reach zero carbon emission by 2050. On the other hand, current energy demand mainly depends on fossil fuels, and is projected to rise by 1% per year until 2040 [2]. In the tension between global energy demand and carbon reduction promises, there exists a widening gap between rhetoric and action [3], and a significant transformation in the energy sector, with extra technical and non-technical efforts, is required to achieve carbon neutrality.

Among various renewable energy resources, wave energy shows great potential in bridging the gap between the rhetoric of carbon reduction and the increasing energy demand, being a relatively untapped resource, with the global wave resource in the range 1-10 TW. However, the exact global estimate of extractable wave power is debatable [4]. The theoretical

#### Abstract

Although wave energy prototypes have been proposed for more than 100 years, they have still not reached full commercialisation. The reasons for this are varied, but include the diversity of device operating principles, the variety of onshore/nearshore/offshore deployment possibilities, the diversity of the wave climate at various potential wave energy sites, and the consequent lack of convergence in technology and consensus. This distributed effort has, in turn, lead to a slow rate of progression up the learning curve, with a significant number of wave energy company liquidations and technical setbacks dampening investor confidence. Although a number of reviews on wave energy technology are already in the published literature, such a dynamic environment merits an up-to-date analysis and this review examines the wave energy landscape from a technological, research and commercial perspective.

> estimate of global wave power is about 32,000 TWh/year (with a mean power of 3.65 TW) [4]. In terms of the usable wave power resource, excluding areas with wave power level < 5 kW/m, the global estimate is around 3 TW [5], while the mean wave power experienced by global oceanic coastlines is about 2.11 TW [6]. The assessment method and data in [5] are used by the Ocean Energy Systems (OES) and the International Renewable Energy Agency, with an estimate of 29,500 TWh/year [7, 8], which exceeds global electricity consumption in 2018, around 22,315 TWh with two-thirds mix from fossil fuels [9]. Together with other renewable resources, wave energy can play an import role in satisfying both the requirements of carbon emission reduction, and energy supply increase. Thus, OES member countries plan to achieve over 300 GW of installed wave/tidal capacity, create 680,000 direct jobs and save 500 Mt of carbon emission by 2050 [7].

> Compared with other renewable resources, especially solar and wind power, the advantages of wave power are multiple: (i) Wave power is characterised by a high-energy density, over 10 times that of wind and solar power [10]. (ii) Wave power has a high availability, up to 90%, while the availability of wind and solar is generally in the range 20-30% [11]. (iii) Wave energy technology has little impact on the environment [12, 13]. (iv)

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

<sup>© 2021</sup> The Authors. IET Renewable Power Generation published by John Wiley & Sons Ltd on behalf of The Institution of Engineering and Technology



FIGURE 1 Possible pathway from WEC R&D activities to commercial deployment, with (a) R&D foci, (b) industry-academia-government collaboration, (b) commercialisation phases, and (d) critical development factors

Wave energy output can also be integrated with existing wind or solar power plants as a complementary resource for smoothing power output and reducing variability [14–19]. (v) Wave power is more predictable [20, 21], giving more flexibility for regional or national power management, and planning.

Despite the enormous potential of wave power, currently active wave capacity is as small as 2.31 MW [8, 22], and these operating wave energy projects are focused on research and demonstration. Currently, wave energy technology is at its 'infant' age, and there is no fully commercial scale wave energy converter (WEC) farm in operation, even though hundreds of WECs have been developed [23]. Crucially, there still exist several technical and non-technical challenges: (i) Technically, it is difficult to generate electricity from low-frequency (0.1 Hz, i.e., low velocity) oscillating motion and large force (1 MN). This requires extremely reliable structures and power take-off (PTO) systems and, consequently, high capital expenditure (CapEx). (ii) WECs operate in an offshore environment, with high installation, operation and maintenance costs. Thus, the operating expenditure (OpEx) is relatively large. (iii) The wave power resource varies on both a wave-by-wave, hour-by-hour, and siteby-site manner, in terms of wave frequency, height, direction, spectrum and power level, resulting in disparate WEC concepts without any convergence, diluting the efforts of research and development (R&D) and commercialisation. (iv) Extreme sea conditions occur from time to time, and the possibility of structural failure and device loss is relatively high. This adds extra risk for the finance sector to invest in WEC technology. (v) Currently, WEC technology is characterised by low maturity, high uncertainty and risk, and requires significant initial capital, which further discourages private investors. That is, diminishing private and public investments has been playing the most important recent role in advancing WEC technology by stimulating R&D activities.

In general, current WEC technologies or devices have not yet demonstrated their capability to harness enough wave energy at a low enough cost at commercial scale. Based on simplistic estimates of the levelised cost of energy (LCoE), some early stage WEC concepts, for example, the M4 device [24, 25], have showed their possibility to achieve a low LCoE for some specific installation sites. Further, geometric optimisation can improve WEC's hydrodynamic performance, in terms of power capture in moderate waves and survivability in extreme waves. On the other hand, sophisticated control approaches can significantly improve power capture, while marginally increasing the CapEx and, hence, dramatically reduce the LCoE [26]. However, WEC hydrodynamics and control are inherently and non-linearly coupled [27, 28], and a co-design approach is needed.

Current R&D activities mainly focus on wave resource assessment, WEC concept developing, hydrodynamic modelling, PTO innovation and control design. As shown in Figure 1(a), the topics in the inner ring are well studied, and plenty of reviews have summarised the state-of-the-art of wave resource assessment [16, 29–31], WEC technology [11, 32–38], modelling [38–47], PTO [11, 36, 48–50], and control [51–55]. The R&D topics in the middle and outer rings in Figure 1(a) are not fully understood yet. There are a few surveys summarising WEC survivability [56], performance [57], economic characteristics [58, 59], mooring [60] and shape optimisation [61, 62]. However, only a few studies aim to investigate critical development factors, as shown in Figure 1(c) and (d), for successful commercialisation of WEC technology at each phase [63–66].

In contrast to the aforementioned reviews, this review aims to discuss potential pathway of WEC commercialisation, from lab to market, by (i) summarising R&D activities in wave resource assessment, PTO innovation, WEC modelling and control, (ii) reviewing ongoing pre-commercial WEC demonstration projects, (iii) identifying potential market opportunities, including the utility market for electricity and niche markets related to ocean applications, and (iv) discussing industryacademia-government (IAG) collaboration to improve some critical development factors for bridging the valley of death (VoD) between R&D and commercialisation activities. WEC technology commercialisation relies not only on technical readiness level (TRL), but also on technical performance level (TPL), which attempts to measure the potential economic performance

7521424, 2021, 14, Downloaded from https://

5rary.wiley.com/doi/10.1049/rpg2.12302 by Battelle Memorial Institute, Wiley Online Library on [29/06/2023]. See the Terms

ons) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons



FIGURE 2 Wave from deep water to shallow water

of a wave energy device/project, and some external development factors, for example, investment environment, market data and national incentives. As the LCoE of WEC technology is still too high to compete with other renewable energy technologies, revenue and capital support from public sectors remains crucial [8]. Thus, public or government-related sectors play an important role in bringing together researchers and investors through support programs, market incentives, and regional policy and legislation, to form a solid IAG collaboration, as shown in Figure 1(b).

The reminder of the paper is organised as follows: Section 2 summarises the basic foundations of ocean waves and wave resource assessment, while Section 3 investigates various WEC concepts, classification, and modelling methods. Section 4 summarises the development of PTO systems and control strategies, with Section 5 examining possible development trajectories of a WEC prototype or project. Section 6 discusses historical and commercial efforts devoted to wave energy technology. Section 7 summaries potential market opportunists for WEC technology, while Section 8 identifies some key factors and incentives for commercialising WEC technology. Finally, some concluding remarks and future perspectives are drawn in Section 9.

### 2 | QUANTIFYING THE WAVE ENERGY RESOURCE

In ocean observation, the wave height H and period T can be directly measured. A simple illustration of wave propagating from deep water to shallow water is given in Figure 2. In Figure 2, h and  $\lambda$  are the water depth and wavelength, respectively. The shallow and deep waters are defined by  $h \leq \frac{\lambda}{20}$  and  $h \geq \frac{\lambda}{2}$ , respectively. As water depth decreases, the shallow water effect reshapes the wave profile, which may result in non-linear waves, wave breaking and energy loss [10, 67]. However, WECs normally operate in moderate sea states with  $H \ll \lambda$ . Thus, linear wave theory is normally valid and is applied in this section, with an overview of regular and irregular waves introduced with specific foci on quantifying the wave resource, its variability, and predictability.

#### 2.1 Regular waves and wave power

Swell waves, characterised by narrow bandwidth and relatively low frequency, are of primary interest for wave energy harvesting, and can be approximated by regular waves. In addition, calculations involving regular waves are relatively simple and can be used as a starting point for WEC R&D activities, particularly at low TRLs. For a regular (monochromatic) wave, the free surface elevation  $\eta$ , in Figure 2, can be written as

$$\eta(x,t) = \frac{H}{2}\cos(\omega t - kx + \varphi), \qquad (1)$$

where  $\omega = \frac{2\pi}{T}$ ,  $k = \frac{2\pi}{\lambda}$  and  $\varphi$  are the wave frequency, wave number and initial wave phase, respectively. The wavelength can be determined by

$$\lambda = \frac{gT^2}{2\pi} \tanh(\frac{2\pi b}{\lambda}), \qquad (2)$$

where g is the gravity constant. Alternatively, Equation (2) is generally rewritten as the dispersion relation, given as  $\omega^2 = gk \tanh(kb)$ . The group velocity of wave propagation in Figure 2 is expressed as

$$c_{\rm g} = \frac{\lambda}{2T} \left[ 1 + \frac{2kb}{\sinh(2kb)} \right]. \tag{3}$$

The potential and kinetic wave energy per unit horizontal area are

$$E_{\rm p} = E_{\rm k} = \frac{\rho g}{2} \overline{\eta^2(x,t)}.$$
 (4)

For regular waves,  $\overline{\eta^2(x,t)} = \frac{H^2}{8}$  holds. Thus, total wave energy per horizontal area is given as

$$E = E_{\rm p} + E_{\rm k} = \frac{\rho_g H^2}{8}.$$
 (5)

Hence, the wave power per unit width of the wave front,  $J_r$ , also called wave-energy transport [68] or wave power level (used hereafter), is given as

$$J_{\rm r} = c_{\rm g} E = \frac{\rho g H^2}{8} \frac{g T}{4\pi} \tanh(kb) \left[ 1 + \frac{2kb}{\sinh(2kb)} \right], \qquad (6)$$

Based on the deep-water assumption, the wave power level can be further simplified as  $J_r \approx \frac{\rho g^2}{32\pi} H^2 T$ .

### 2.2 | Irregular waves and wave power

In general, real waves are random and irregular, and can be approximated by the superposition of a group of sinusoidal waves as

$$\eta(x,t) = \sum_{i=1}^{N} \frac{H_i}{2} \cos(\omega_i t - k_i x + \varphi_i), \qquad (7)$$

where  $H_i$ ,  $\omega_i$ ,  $k_i$  and  $\varphi_i$  are the wave height, frequency, wave number and initial phase of the *i*th sinusoidal wave of Ncomponents, respectively. Based on linear wave theory, Equations (2)–(4) still hold. Thus, total wave energy power per horizontal area for irregular waves can be written as

$$E = \rho_g \overline{\eta^2(x,t)} = \rho_g \int_0^\infty S(\omega) \, \mathrm{d}\omega, \qquad (8)$$

where  $S(\omega)$  is the wave energy spectrum. Based on time-domain ocean observations, the wave spectrum can be estimated via FFT or spectrum estimation methods.

A variety of wave spectral models have been discussed in [67, 69], based on their application scenarios, pros and cons, of which the notable ones are the Pierson–Moskowitz (PM) [70] and joint North Sea wave project spectral models [71]. Here, the PM spectrum, generally used to describe fully developed wind waves, is taken as an example, given as

$$S(\boldsymbol{\omega}) = \frac{5H_{\rm s}^2}{16} \frac{\omega_{\rm p}^4}{\omega^5} \exp\left(-\frac{5\omega_{\rm p}^4}{4\omega^4}\right),\tag{9}$$

where  $H_{\rm s}$ ,  $\omega_{\rm p} = \frac{2\pi}{T_{\rm p}}$  and  $T_{\rm p}$  are the significant wave height, peak frequency and peak period, respectively. For a given wave spectrum, the significant wave height and energy period,  $T_{\rm e}$ , can be estimated from spectral moments as

$$H_{\rm s} = 4\sqrt{m_0},\tag{10}$$

$$T_{\rm c} = \frac{m_{-1}}{m_0},$$
 (11)

where  $m_i$  is the *i*th moment of the spectrum, given as

$$m_i = \int_0^\infty \omega^i \mathcal{S}(\omega) \, \mathrm{d}\omega. \tag{12}$$

Thus, the wave power level for irregular waves can be written as

$$J = \frac{\rho g^2}{64\pi} T_{\rm e} H_{\rm s}^2. \tag{13}$$

# 2.3 | The global wave power resource and spatial variability

To estimate the wave resource over a large area, numerical wave models are generally used, of which the notable ones are the wave model, wavewatch 3, simulating waves nearshore, MIKE21-SW and TOMAWAC models, with their limitations and application scenarios discussed in [30, 31]. To achieve accurate wave resource assessment, observed wave data, at a set of discrete spatial points, are used to calibrate the models. Although the exact global estimate of extractable wave power is

debatable, depending on assessment method, wave model, and temporal and spatial resolution, a small set of studies conclude that the applicable wave power in the world is about 3 TW by excluding areas of J < 5 kW/m [5, 7, 8]. Considering the area with 30 nautical miles to the coastline, extractable wave power decreases to 2.11 TW [6], and decreases further to 1.85 TW (approximating 16,000 TWh/year), when wave direction and coastline alignment are considered [4].

The wave power resource is evenly distributed between the Southern and Northern Hemispheres, as shown in Figure 3(a), but is concentrated within 30-60 degrees of latitude. Thus, latitude is one main factor affecting the spatial variability of the wave power resource. One typical example is the wave power resource along the Chilean coast, as shown in Figure 3(b), with the wave power level increasing from 20 to 100 kW/m, as the latitude increases from 15°S to 55°S. It also shows that water depth has some influence on the wave power level. As waves propagate to the coastline, the shallow water effect causes energy loss. Consequently, spatial variability has a significant influence on WEC performance [72, 74, 75]. As shown in Figure 3(c), the capacity factor of 3 WECs increases, as the latitude and wave power level increase. When wave power level is low, WECs should be scaled down accordingly to improve their performance [72, 75].

# 2.4 | Temporal variability and predictability of wave power

Wave power is characterised by significant temporal variability, ranging from seconds to decades. Such high temporal variability is one reason for the diversity of WEC concepts, and points to a required focus on WEC optimisation, PTO, control, survivability, power prediction, and management. Temporal variability can be classified into short-, medium- and long-term variations.

Short-term variation is characterised by irregularity in height, period and direction, varying from seconds to minutes. As the WEC control problem is typically non-causal, short-term prediction of wave elevation or excitation force is required, and prediction requirements for real-time control are investigated in [76]. Several prediction methods are discussed in [76–83], including the AR, ARMA, NARX and Bayesian learning methods. In addition, short-term variation results in a highly varying instantaneous wave power and a high peak-to-average power ratio, and extra design effort is required to smooth WEC harvested power, for example, PTO systems with accumulators/flywheels, to smooth high-frequency power variation.

Medium-term variation is represented by a change in wave spectrum or sea states, on an hourly or/and daily basis. Such variation may challenge the power management system of WEC farms, and accurate wave prediction over 1–72 h is required for power planning [16], and WEC installation and maintenance [83]. Compared with other renewable resources, wave power has an advantage in predictability [20, 21, 84], and the significant wave height can be accurately predicted in advance by a couple of days [16, 83, 85]. In addition, forming WEC arrays, or



itions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Common

**FIGURE 3** (a) Global wave resource distribution [6], (b) spatial variability along the Chilean coast, data from [72], (c) influence of wave power level variation and water depth on WEC capacity factor, data from [72], (d) monthly variability in the Northern and Southern Hemispheres, data from [6], and (e) seasonal and annual variability at the Atlantic Marine Energy Test Site, data from [73]

integrating WECs with wind turbines, can smooth power output to overcome medium-term variation [14].

Long-term variation concerns intra-annual and inter-annual variability of the wave power resource [4, 86–88]. Intra-annual variability includes monthly and seasonal variability, while the inter-annual variability refers to wave power variation over decades. In general, wave power is high in winter and low in summer, as shown in Figure 3(d) and (e). Inter-annual variability has a significant influence on lifetime performance of WEC farms and, thus, should be considered when determining deployment sites and design capacity ratings [87, 89, 90].

For instance, the wave power on the west coast of Ireland has seen a significant increase in the 20th century, which shows a power surplus of 15% within the lifespan of a point absorber (PA) or an oscillating wave surge converter (OWSC) [87, 91]. However, extreme events also increase, requiring more focus on WEC survivability [87]. In addition, increase in off-limit events ( $H_s \ge 5m$ ) can significantly reduce the capture width ratio of an OWSC in the Irish sea, up to a level of 20% [92]. Thus, long-term trends of wave climate should be considered for commercial planning [29, 73]. However, long-term variability can, in general, be only hindcasted rather than forecasted [83]. 17521424, 2021, 14, Downloaded from https

library.wiley.com/doi/10.1049/rpg2.12302 by Battelle Memorial Institute, Wiley Online Library on [29/06/2023]. See the Terms and Conditions (https://



FIGURE 4 The average wave power level and the coefficient of variation for various typical sites, data from [94]. In this figure, the mean value for these sites is marked by the red dot; 'S', 'N', 'W', 'C' represent 'south', 'north', 'west' and 'central', respectively

Statistical methods are generally used to quantify the temporal variability of wave resource. One simple measure is the coefficient of variation (CoV) [93], given as

$$C_{\theta}V(J) = \frac{\sqrt{\left(J(t) - \overline{J(t)}\right)^2}}{\overline{J(t)}},\tag{14}$$

where *t* is the time interval used for computing the wave power level *J*, ranging from 30 min to decades. Thus, the CoV can be used to describe medium- and long-term variability. Based on the CoV definition in Equation (14), a new world map for wave power with long-term variability is studied in [94], and shown in Figure 4. For various selected sites, the average value of (CoV, J) is marked by the red dot, which also divides Figure 4 into four regions. The sites in the green region (top left corner) are ideal for WEC installation, as the wave power level is high and the CoV is low.

Intra-annual variability can be quantified by the monthly variation index (MVI) and seasonal variation index (SVI) [93], while inter-annual variability is represented by the annual variation index (AVI) [86], as

$$MVI = \frac{J_{\rm M,max} - J_{\rm M,min}}{J_{\rm Y}},$$
(15)

$$SVI = \frac{J_{\rm S,max} - J_{\rm S,min}}{J_{\rm Y}},\tag{16}$$

$$AVI = \frac{J_{Y,max} - J_{Y,min}}{\overline{J_Y}},$$
(17)

where the suffixes M, S, and Y represent month, season and year, respectively, while the suffixes max and min represent maximum and minimum values, respectively. These measures are generally adapted to quantify temporal variability of wave power resource globally, regionally, and/or locally [4, 14, 16, 20, 29, 73, 86].

# 2.5 | Influence of wave climate on commercialisation

For commercialising WEC technology, the first step is to select a deployment site, mainly according to annual mean wave power level and temporal variability. Sites with a high wave power level but low variability are preferred, and WECs should be selected accordingly. Variation in wave climate is a strong cost driver in both CapEx and OpEx [94–96]. Short- and medium-term variability can be handled by real-time control and power management, along with wave climate prediction. However, long-term variability is difficult to forecast but can significantly affect the lifetime performance of WEC farms.

In addition, extreme wave conditions, characterised by maximum wave height and storm occurrence, have significant influence on accessibility and availability of wave power, and survivability of WEC devices. More R&D activities are needed to improve WEC survivability in extreme waves. To ease the installation and operation of WEC farms, other key factors should be considered, including water depth, distance to the coast, wind and tidal climate, existing infrastructure, and environmental and spatial constraints [89, 97].

## 3 | WAVE ENERGY TECHNOLOGY PRINCIPLES

This section gives an overview of working principles for various WEC concepts and their classification, followed by an overview of hydrodynamic modelling of WECs, based on these principles.

# 3.1 | Wave energy conversion concepts and classification

A WEC device converts the kinetic and/or potential energy contained in moving waves to useful energy (mainly electricity), comprising a set of floating or submerged bodies, a PTO unit, a control system, power electronics and other accessories. Since wave energy conversion concepts diverge, with over 1000 devices reported [10], there is no unique categorisation method to cover all possible WEC systems. In general, WECs can be classified according to their deployment locations, working principles, operation modes and device geometries [32, 34, 35, 38]. In this study, the classification method detailed in [35] is adapted and shown in Figure 5. In Figure 5, WECs are classified into three types, including oscillating water columns (OWCs), wave activated bodies and overtopping devices. For each type, the exemplified prototypes are pre-commercial and have been tested in the open ocean.



FIGURE 5 Classification of WEC devices, inspired by [35]

An OWC utilises a hollow structure with an open inlet below the still water level to trap air in its chamber above the inner free-surface; wave action alternately compresses and decompresses the trapped air, which forces air to flow through a turbine coupled to a generator [36], principally using the kinetic wave energy. As listed in Figure 5, OWCs can be further catalogued into two subclasses: (i) fixed OWCs, for example, the *Pico* and *LIMPET* devices, and (ii) floating OWCs, for example, Masuda's navigation buoy, and the *Spar-buoy* OWC. A comprehensive review, with a specific focus on OWCs and their PTO systems, is summarised in [36].

Overtopping devices are exemplified by fixed prototypes such as the *TAPCHAN* and *OBREC* devices, or the floating *Wave Dragon (WD)* device. Overtopping WECs mainly use potential wave energy, with electricity generated via somewhat conventional unidirectional (low head) hydro-turbines.

Most R&D activities focus on wave-activated WEC concepts, which can make use of the potential or/and kinetic wave energy to generate electricity [11, 35, 36, 62]. Wave-activated WECs can be further classified as (i) floating or submerged subclasses, according to wave-WEC interaction, (ii) rotating or translating subtypes according to the essential degrees of freedom (DoFs) exploited, or (iii) PAs, attenuators and terminators according to WEC geometry, with respect to wavelength and propagating direction. PAs refer to WEC devices whose characteristic dimensions are much smaller than the incident wavelength. PAs may operate in heave, pitch or multiple DoFs, and can be situated nearshore or offshore. Attenuators are floating WEC devices, oriented parallel to the wave direction, usually composed of multiple floating bodies connected by hinged joints, with relative motion between two connected bodies used to generate electricity via PTO systems. Terminators are oriented perpendicular to the wave direction, typically including duck-like devices, or OWSCs. Some typical waveactivated type WECs, at pre-commercial scales, are listed in Figure 5.

#### 3.2 | Hydrodynamic modelling

Mathematical modelling of WEC dynamics gives a foundation for WEC R&D activities, and an accurate mathematical model



FIGURE 6 Governing equations for CFD and PFT

can be used for evaluating WEC performance, developing control strategies and optimising system design according to a specific wave climate. In the literature, a considerable amount of R&D activities have been devoted to WEC hydrodynamics, summarised in [38, 40, 41, 44–47, 98]. One main challenge of WEC modelling is how to represent the wave-structure interaction (WSI) in a proper way. Typically, the motion of a WEC structure is governed by Newton's 2nd Law, as

$$\boldsymbol{M}\boldsymbol{\xi}(t) = \boldsymbol{f}_{\rm h}(t) + \boldsymbol{f}_{\rm g}(t) + \boldsymbol{f}_{\rm pto}(t) + \boldsymbol{f}_{\rm ext}(t), \qquad (18)$$

where  $\boldsymbol{M}$  is the inertial matrix and  $\boldsymbol{\xi}$  is the displacement of the WEC.  $\boldsymbol{f}_{\rm h}, \boldsymbol{f}_{\rm g}, \boldsymbol{f}_{\rm pto}$  and  $\boldsymbol{f}_{\rm ext}$  are the hydrodynamic, gravity, PTO (or control), and external forces, respectively. There is no unique representation of  $\boldsymbol{f}_{\rm ext}$ , but it may contain mooring and other potentially non-linear forces, for example, end-stop force [99–101]. Hydrodynamic fluid/body force can be computed as the integral of the pressure p on the wetted surface S, given as

$$\boldsymbol{f}_{\rm h} = -\iint \,_{\boldsymbol{S}} \boldsymbol{p} \, \boldsymbol{n} \, \mathrm{d}\boldsymbol{S},\tag{19}$$

where n is the normal vector on the wetted surface. Thus, the key of hydrodynamic modelling is to compute the pressure p in the fluid. Fluid dynamics is governed by the Navier– Stokes equations (NSEs), which cannot be solved analytically, and computational fluid dynamics (CFD) methods are generally applied to obtain a numerical solution. By assuming an ideal fluid (incompressible, inviscid and irrotational), the NSEs can be simplified to the Laplace and Bernoulli equations in Figure 6, and then solved using potential flow theory (PFT). With the pressure p obtained via CFD or PFT methods, the hydrodynamic force in Equation (19) can be computed.

CFD methods have been widely applied to provide highfidelity numerical solutions to WSI, which can be classified into Eulerian and Lagrangian methods [44, 46]. Eulerian methods discretise the fluid into mesh elements, while Lagrangian approaches discretise the fluid as a set of particles. Eulerianbased CFD packages, for example, ANSYS Fluent, CFX, FLOW-3D, Star-CD/CCM+, and OpenFOAM, are generally used for modelling WEC dynamics, since they can handle all kinds of non-linear WEC hydrodynamics, for example, nonlinear waves, turbulence, overtopping and slamming. On the other hand, Lagrangian-based smooth particle hydrodynamics (SPH) methods, for example, DualSPHysics [102], show advantages in automatic conservation of mass, and simplification of surface tracking, particularly suitable for extreme wave events, for example, wave breaking. However, SPH methods are not yet fully validated.

While providing high-fidelity modelling results, the computational cost of CFD approaches are expensive, with typically  $10^4-10^5$  seconds of computation (real) time for 1 s of simulation time [27, 103], significantly depending on CFD setups. By assuming sea water is ideal, fully non-linear potential flow (FNPF) theory can be used to solve the Laplace and non-linear Bernoulli equations in Figure 6, and compute the velocity potential function  $\phi$  [44]. Thus, the pressure can be computed by

$$p = -\rho_{g\chi} - \rho \frac{\partial \phi}{\partial t} - \frac{\rho}{2} (\nabla \phi)^2.$$
 (20)

Currently, only a few software packages are based on the FNPF theory, for example, the high-order boundary element method of Ning *et al.* [104, 105]. Further assuming the wave height is much smaller than the wavelength ( $H \ll \lambda$ ) and the device motion is small, linear potential flow (LPF) theory can be applied to solve the Laplace and linear Bernoulli equations in Figure 6. Thus, the total potential function can be divided into incident, diffracted and radiated components, as

$$\boldsymbol{\phi} = \boldsymbol{\phi}_{\mathrm{i}} + \boldsymbol{\phi}_{\mathrm{d}} + \boldsymbol{\phi}_{\mathrm{r}}.$$
 (21)

For linear incident waves, an analytical solution of  $\phi_i$  generally exists. However, analytical solutions for  $\phi_d$  and  $\phi_r$  only exist for some simple WEC structures, for example, spheres and cylinders [106–108]. For arbitrary WEC geometries, meshbased boundary element methods (BEMs) are generally used to obtain numerical approximations of  $\phi_d$  and  $\phi_r$ . Common BEM solvers include WAMIT, NEMOH, AQWA, AQUA+ and WADAM in the frequency domain, and ACHIL3D in the time domain [47]. Substituting  $\phi_i$ ,  $\phi_d$  and  $\phi_r$  in Equations (20) and (21) and omitting the quadratic term in Equation (20), the pressure p is obtained, to allow the hydrodynamic force in Equation (19) to be computed.

In the time domain (TD), Equation (18) can be rewritten as Cummins' equation [109], given as

$$\boldsymbol{M}\boldsymbol{\ddot{\xi}}(t) = \boldsymbol{k}_{\mathrm{c}}(\tau) * \boldsymbol{\eta}(t) + \left[-\boldsymbol{M}_{\infty}\boldsymbol{\ddot{\xi}}(t) - \boldsymbol{k}_{\mathrm{r}}(\tau) * \boldsymbol{\dot{\xi}}(t)\right] - \boldsymbol{K}\boldsymbol{\xi}(t) + \boldsymbol{f}_{\mathrm{pto}}(t) + \boldsymbol{f}_{\mathrm{ext}}(t), \qquad (22)$$

where  $\mathbf{f}_{e}(t) = \mathbf{k}_{e}(\tau) * \boldsymbol{\eta}(t)$  is the excitation force, as a sum of the Froude-Krylov (FK) and the diffraction forces,  $\mathbf{f}_{r}(t) = -\mathbf{M}_{\infty}\ddot{\mathbf{\xi}}(t) - \mathbf{k}_{r}(\tau) * \dot{\mathbf{\xi}}(t)$  is the radiation force, and  $\mathbf{f}_{hs}(t) = -\mathbf{K}\mathbf{\xi}(t)$  is the hydrostatic force representing the balance between the gravity and buoyancy forces.  $\mathbf{k}_{e}(\tau), \mathbf{M}_{\infty}, \mathbf{K}$  and  $\mathbf{k}_{r}(\tau)$  are the excitation kernel function, added mass at infinite frequency, restoring stiffness, and radiation kernel function, respectively. Alternatively, WEC dynamics can be written in the frequency domain (FD) as

$$\{-\omega^{2}[M + M_{a}(\omega)] + j\omega B(\omega) + K\}\Xi(\omega)$$
$$= K_{e}(\omega)\mathcal{A}(\omega) + F_{pto}(\omega) + F_{ext}(\omega), \qquad (23)$$

where  $\Xi(\omega)$ ,  $A(\omega)$ ,  $F_{e}(\omega) = k_{e}(\omega)A(\omega)$ ,  $F_{r}(\omega) = [\omega^{2}M_{a} - j\omega B]\Xi(\omega)$ ,  $F_{pto}(\omega)$  and  $F_{ext}(\omega)$  are the frequency-domain representations of  $\xi(t)$ ,  $\eta(t)$ ,  $f_{e}(t)$ ,  $f_{r}(t)$ ,  $f_{pto}(t)$  and  $f_{ext}(t)$ , respectively.  $K_{e}(\omega)$ ,  $M_{a}(\omega)$  and  $B(\omega)$  are the excitation frequency-response function, added mass and radiation damping, respectively.

The parameters,  $K_e(\omega)$ ,  $k_e(\tau)$ ,  $M_{\infty}$ ,  $M_a(\omega)$ ,  $k_r(\tau)$ ,  $B(\omega)$ and K, can be obtained from the aforementioned BEM codes. The TD and FD models in Equations (22) and (23) can be connected according to Ogilvie's relations [110]. The TD and FD models can accurately depict WEC dynamics if the body motion is small. However, this is not always the case, especially when power maximisation control is applied to exaggerate WEC motion. In this case, some critical non-linear forces cannot be neglected any more, and hybrid modelling methods are generally applied to add some critical non-linear terms as treatments to  $f_{ext}(t)$  in Equation (22).

By superimposing critical non-linear terms, a higher modelling fidelity can be achieved without a significant increase in computational cost. However, dominant non-linear factors depend significantly on specific WEC concepts, structure sizes, control strategies and application scenarios, and should be carefully considered on a case-by-case basis [42, 45, 111]. Depending on the additional non-linear term to  $f_{ext}(t)$ , hybrid modelling methods are divided into four types [47], including the bodyexact, weak-scatterer, viscosity and mixed treatments.

The body-exact treatment considers instantaneous body motion when computing the FK, diffraction, radiation and restoring forces, covering large WEC motion. A critical aspect is the non-linear FK force, which has a large influence on WEC hydrodynamics [112–114]. The weak-scatterer treatment considers the instantaneous free surface while the wetted surface boundary condition is linearised at its mean value, allowing highorder potential functions for computing the pressure and hydrostatic force in Equations (19) and (20).

Fluid viscous effects can be represented by adding a quadratic drag term to Cummins' equation, via a Morison term [115], given as

$$\boldsymbol{f}_{v} = \frac{1}{2} \rho C_{d} A_{s} (\boldsymbol{u} - \boldsymbol{v}) |\boldsymbol{u} - \boldsymbol{v}|, \qquad (24)$$

where  $A_s$  is the section area normal to the relative flow direction,  $\boldsymbol{u}$  is the water particle velocity, and  $\boldsymbol{v} = \dot{\boldsymbol{\xi}}$  represents the velocity vector of the WEC body.  $C_d$  is the viscous coefficient, depending on the Keulegan–Carpenter number, the Reynolds number and the roughness number [116]. Such a viscosity treatment can improve modelling accuracy significantly, especially for large relative fluid/body velocities. However, it is non-trivial to determine  $C_d$ , and a wide range of wave conditions should be tested to obtain a consistent value [117–121]. To improve power capture, viscous effects should be minimised by geometric optimisation. For example, the M4 device utilises hemi-spherical bases to minimise viscous losses [24, 25].

The mixed treatment combines the viscosity representation with the body-exact or the weak-scatter treatments, leading to body-exact-viscosity or the weak-scatter-viscosity models. The former is more generally used for modelling WEC hydrodynamics, as controlled WECs are expected to oscillate with a large motion, even in moderate sea states [27, 47]. So, the bodyexact-viscosity treatment is useful for modelling WEC dynamics in normal operation mode, where the modelling fidelity of a heaving PA, considering non-linear FK and viscous forces, can approach CFD results, with significantly lower computational cost [103].

Although the TD model in Equation (22) shows high flexibility in handling non-linear treatments, the excitation and radiation force convolution terms are not efficient for WEC R&D activities. Thus, system identification techniques are generally used to approximate the radiation convolution terms by finiteorder parametrised models, for example, state-space models [122–127]. As the excitation kernel function  $\mathbf{k}_{e}(\tau)$  is generally non-causal, extra effort is required to represent the excitation force [79, 81, 128–132]. More generally, system identification methods are applied to derive compact linear and non-linear models directly from CFD, or experimental, data [133–137].

# 3.3 | Influence of WEC technology on commercialisation

Dilution of R&D effort across many WEC concepts may be one main reason for currently low TRL and immaturity of WEC technology. To advance the convergence of WEC technology and concentration of R&D efforts, a common consensus on performance metrics for ocean energy technology is developing via international collaboration [66]. Hopefully, such a framework will accelerate the convergence of WEC technology, and consequently improve its TRL, maturity and commercialisation potential.

Hydrodynamic modelling of WECs has been well studied, but mainly for operational mode in moderate sea states. However, several WEC structural failures in storm conditions are reported in [138], even causing complete device loss. Recently, a few studies investigated WEC dynamics in extreme waves via tank testing [24, 25] or CFD simulation [56, 139, 140]. However, more research focus and effort on WEC dynamics in extreme waves are required to evaluate WEC survivability, reducing the risk of WEC commercialisation. Although there are some studies on the optimisation of WEC farm layout [141–145], only linear hydrodynamic models of WEC farm are typically used. For full commercial-scale WEC farms with tens of WEC devices, the interaction between WECs is not yet fully understood.

### 4 | POWER TAKE-OFF SYSTEM AND CONTROL

The PTO system is one key component of a WEC device, which transforms the mechanical power from the WEC motion to electricity. The PTO system has its own dynamics which, allied to those of the floater hydrodynamics, determines the overall frequency response characteristics of the system, which needs to be tuned to the relevant sea state via the control system. Conse-



FIGURE 7 PTO systems for wave energy technology [160], courtesy of Professor Amélie Têtu

quently, a reliable and robust PTO, together with an appropriate control strategy, will improve commercialisation potential.

#### 4.1 | Power take-off system

Various PTO systems are illustrated in Figure 7, including: (i) hydraulic PTOs, applicable to many kinds of WEC concept [49, 50, 146–152], (ii) air turbines for OWCs [36, 153], (iii) hydro turbines for overtopping devices [154], (iv) mechanical rectifiers [155, 156], and (v) direct-drive generators [157–159]. In general, these PTO systems are well modelled and market available, often derived from other application areas.

In general, PTOs cannot directly integrated to WEC devices, due to the following technical challenges: (i) PTOs are generally optimised to operate efficiently at a high unidirectional speed. However, WEC motion is typical slow, with mechanical rectifiers and mechanical or hydraulic gearing used to produce higher speed unidirectional motion. (ii) As discussed in Section 2, waves can have high temporal variability, making it difficult to determine the rated specifications for PTO components. In addition, short-term variability induces a high peakto-average power ratio, which may lead to occasional overrated conditions. Thus, PTOs which can operate efficiently over a wide range of sea states are required. (iii) Extreme sea state occurs occasionally, exceeding the PTO physical constraints, for example, maximum stroke, velocity, force and power, so PTO decoupling mechanisms are required.

Current WEC modelling tends to assume that PTOs in Figure 7 are ideal, simplified the model to a mass-spring-damper system, or neglecting some important non-linear factors, for example, hysteresis effect [148], dead-zone, saturation, friction [151], and load effects [161]. This may lead to incorrect design decisions regarding the PTO and control systems. For high-fidelity PTO modelling, both their dynamics and efficiency variations with load should be considered. In general, the average efficiency of a non-ideal PTO deceases as the reactance/resistance ratio increases [161], and the average PTO efficiency decreases dramatically when the load diverges from its rated value. A large amount of energy will also be dissipated by non-ideal PTOs, in terms of hydraulic leakage [152], mechanical loss [151, 152, 155], or copper loss [162–164]. Thus, non-linear and non-ideal PTO factors should be modelled and then considered at the control design stage.

Since PTOs are naturally non-ideal and non-linear, it is more realistic to integrate a non-linear PTO model with a nonlinear WSI model to form a high-fidelity wave-to-wire (W2W) model for control, optimisation and performance evaluation [43, 45, 54, 165]. However, such a model is complex and expensive in computation. Thus, systematic complexity reduction approaches are required to achieve an acceptable balance between fast computing and model fidelity, discussed in [164, 166, 167].

## 4.2 | Control

Since ocean waves are irregular in amplitude and frequency, and sea states change all the time, control approaches are required by WECs for power maximisation in mild/moderate waves and survivability enhancement in extreme waves. A wide range of WEC control strategies are available, for example, reactive control [168], phase control [68], optimisation-based control [53], adaptive control [169, 170]. This section only discusses some basic concepts of WEC control, major milestones in the literature, and their influence on WEC commercialisation. Detailed control reviews are given in [51, 53–55].

### 4.2.1 | Classical control strategy

The fundamental (classical) study of power maximising control of WEC systems dates back to 1975 [168], which is based on monochromatic waves, linear hydrodynamics and an ideal PTO. To derive optimal conditions, Cummins' equation in Equation (23) can be rewritten as

$$\frac{V(\omega)}{F_{\rm e}(\omega) + F_{\rm pto}(\omega)} = \frac{1}{Z_{\rm i}(\omega)},\tag{25}$$

where  $V(\omega)$  represents the body velocity in the frequency domain.  $Z_i(\omega)$  is the intrinsic impedance of the system, given as

$$\mathbf{Z}_{i}(\boldsymbol{\omega}) = \mathbf{B}(\boldsymbol{\omega}) + j\boldsymbol{\omega} \left[ \mathbf{M} + \mathbf{M}_{a}(\boldsymbol{\omega}) - \mathbf{K}/\boldsymbol{\omega}^{2} \right].$$
(26)

According to the maximum power transfer theorem, the optimal PTO system should satisfy

$$\boldsymbol{Z}_{\text{pto}}(\boldsymbol{\omega}) = \boldsymbol{Z}_{i}^{*}(\boldsymbol{\omega}). \tag{27}$$

Thus, the absorbed power is maximised when the PTO impedance is the complex conjugate of the system intrinsic impedance, so-called complex-conjugate control, or reactive control (RC) [68], with bidirectional power flow required in

7521424, 2021, 14, Downloaded from https:

prary. wiky.com/doi/10.1049/ppg2.12302 by Battelle Memorial Institute, Wiky Online Library on [29/06/2023]. See the Terms and Conditions (https://onli

ibrary.wiley

conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License



**FIGURE 8** (a) Approximate complex conjugate (ACC) and (b) approximate velocity tracking (AVT) control frameworks [55]

Equation (27). With some PTOs only allowing uni-directional power flow, for example, dampers, RC degenerates to so-called passive control (PC), as  $B_{\text{pto}}(\omega) = |Z_i(\omega)|$ .

For RC in Equation (27), the optimal velocity is written as

$$V_{\text{opt}}(\omega) = \frac{F_{\text{c}}(\omega)}{2B(\omega)},$$
 (28)

which indicates that the absorbed power is maximised if the amplitude and phase conditions,

$$|V_{\text{opt}}(\omega)| = \frac{|F_{\text{c}}(\omega)|}{2|B(\omega)|} \text{ and } V_{\text{opt}}(\omega) = \frac{F_{\text{c}}(\omega)}{2B(\omega)},$$

are simultaneously satisfied. For a given frequency, the amplitude condition can be achieved by selecting a suitable damper, while the phase condition can be achieved by latching control (LC) [68] or declutching control (DC) [171].

In addition to strong assumptions of monochromatic wave, linear hydrodynamics and ideal PTO, physical constraints, for example, WEC stroke, cannot be easily handled by classical control strategies and, thus, their value is limited. However, Equations (26)–(28) have established the theoretical foundation for WEC control.

#### 4.2.2 | Modern control strategies

Ocean waves are generally panchromatic, and the optimal control laws in Equations (26)–(28) can be extended to the approximate complex-conjugate (ACC) and approximate velocity tracking (AVT) structures [55], respectively, as shown in Figure 8. The ACC structure, which is optimal only for the predominant wave frequency, does not require any estimation or forecasting of wave excitation force, but cannot handle constraints directly, and hard to improve its robustness to modelling errors [172]. Meanwhile, the AVT framework is more flexible than the ACC one, permitting the incorporation of physical constraints, and is generally used for the majority of optimal control strategies, but requires knowledge of the excitation force, and is significantly more computationally complex. In general, the WEC control problem can be rewritten as an optimal control problem, given as

$$\begin{split} \min_{\boldsymbol{f}_{\text{pto}}} &- \int_{0}^{T} \boldsymbol{f}_{\text{pto}}(t) \, \boldsymbol{v}(t) \, \mathrm{d}t \\ \text{s.t.} \boldsymbol{\xi}(t) \leq \boldsymbol{\xi}_{\text{max}}, \\ \boldsymbol{f}_{\text{pto}}(t) \leq \boldsymbol{f}_{\text{max}}, \end{split} \tag{29}$$

where  $\boldsymbol{\xi}_{max}$  and  $\boldsymbol{f}_{max}$  are the PTO constraints in stroke and force, respectively. Based on this formulation, several optimal control algorithms are summarised in [53]. Among various optimal control approaches, model predictive control (MPC) is widely investigated, for linear hydrodynamics [173, 174], non-linear hydrodynamics [175], non-ideal PTO [162, 176], and even WEC arrays [177].

Numerical optimisation in Equation (29) is time consuming, which may cause some difficulty in real-time implementation. In addition, the WSI is so complex that the convexity of Equation (29) is not generally guaranteed. Further considering physical constraints in WEC stroke and PTO force, the existence of optimal control solutions is also not guaranteed. Once wave excitation exceeds a certain level, there may be no control solution that simultaneously satisfies PTO constraints in force and displacement [55, 178]. Thus, the existence of an optimal solution depends on wave conditions, WSI and the PTO specification. On the other hand, a well-controlled WEC tends to oscillate significantly, resulting in large body motion which, in turn, exaggerates non-linear hydrodynamics [27, 28]. Thus, control should be considered in WSI and PTO modelling, requiring a relatively high-fidelity W2W model, inherently considering non-linear WSI and non-ideal PTO. However, such a model is typically complex, and systematic complexity reduction is inevitable [164, 166, 167].

### 4.2.3 | New trends in control

To advance WEC control implementation, recent R&D efforts have been devoted to robust control, model-free control (MFC), and system complexity reduction.

Since WSI is somewhat of a 'blackbox', modelling error and uncertainty are inevitable. Thus, control strategies have been designed to address the robustness to: modelling error [80, 179–183], external disturbance [184–186], and estimation/prediction error of excitation force [187]. Model sensitivity issue for different WEC control system architectures is analytically and numerically studied in [172], revealing that the AVT architecture is relatively insensitive to inertial and stiffness terms, and generally has superior robustness compared with the ACC framework [172].

MFC is an alternative to robust control, aiming to make WEC performance more immune to model uncertainty, external perturbation and unmodelled dynamics. Among various MFC control methods, some notable examples are extremumseeking algorithms [188], artificial neural networks [189], deep reinforced learning control [190], machine learning [191], leastsquares policy iteration [192] and adaptive control [170]. Some of these MFC methods inherently contain a large number of optimisation iterations, resulting in real-time implementation challenges.

A high-fidelity W2W model, with non-linear WSI and nonideal PTO, is naturally complex, with complexity reduction required for control and optimisation [43, 45, 54, 165]. For realtime implementation, an interesting aspect is to avoid some of the particular problems, including (i) excitation force estimation, (ii) excitation force forecasting, and (iii) numerical optimisation [55].

# 4.3 | Influence of PTO and control on commercialisation

For advancing PTO designs and implementation, R&D activities should use a realistic and non-ideal model for numerical investigations in modelling, performance evaluation, control development and design optimisation. For practical testing, attention should be paid to PTO optimisation with respect to wave climate, to improve its reliability for long-term operation. In addition, a durable PTO should have a decoupling design to survive in extreme waves.

Control plays the most important role in advancing WEC economic performance by reducing LCoE. Although WEC controllers increase CapEx by a small margin, annual energy production can be improved significantly. A typical example is the *SEAREV G21* device [26], where a properly designed control system increases the annual energy production from 730 MWh to 1300 MWh, with the CapEX only increasing from 5 M€ to 5.3 M€. In general, hydrodynamics, PTO dynamics and control are inherently coupled in a non-linear manner, and such a coupling is not yet fully understood [27, 28]. Thus, it is imperative to establish a co-design framework to accurately address such a coupling in its true form.

Supervisory control, to switch WEC system between operation and survival modes according to sea sates, and fault tolerant control, to improve system reliability, are seldom tackled. Several studies have addressed the array control problem [193–197], but are limited to small device numbers and simple array layouts.

### 5 | WEC DEVELOPMENT TRAJECTORIES

To take a high-tech product from lab to market, it is critical to evaluate the maturity of the technology, mainly represented by TRL. However, a successful commercial strategy also requires that the technology is marketable and investable, represented by TPL. Based on a TRL-TPL-matrix, it is possible to find an ideal development trajectory for WEC projects, even at very early development stages. To date, some WEC projects have achieved either high TRL or TPL. However, high TRL and TPL do not naturally indicate successful commercialisation, and many new ventures have failed to bridge the VoD. In this section, the development trajectory of R&D WEC projects will be discussed

TABLE 1 Characteristics of functional readiness and lifecycle readiness for TRLs, adapted from [198]

TRL	Function readiness	Lifecycle readiness
1	Basic principles observed and reported.	Potential uses of technology identified.
2	Technology concept formulated.	Market and purpose of technology identified.
3	Analytical/experimental key function and/or characteristic proof-of-concept (scale < 1:25).	Initial capital cost and power production estimates or targets established.
4	Technology component and/or basic technology subsystem validation in a laboratory environment ( <b>scale &gt; 1:25</b> ).	Preliminary lifecycle design: targets for manufacturable, deployable, operable, and maintainable technology.
5	Technology component and/or basic technology subsystem validation in a relevant environment ( <b>scale &gt; 1:15</b> ).	Supply-chain mobilisation: procurement of subsystem design, installation feasibility studies, cost estimations, etc.
6	Technology system prototype demonstration in a relevant environment ( <b>scale &gt; 1:4</b> ).	Customer interaction: consider customer requirements to inform type design. Inform customer of likely project site constraints.
7	Technology system prototype demonstration in an operational environment ( <b>scale &gt; 1:2</b> ).	Ocean operational readiness: management of ocean scale risks, marine operations, etc.
8	Actual product completed and qualified through test and demonstration ( <b>full scale, 1 device</b> ).	Actual marine operations completed and qualified through test and demonstration.
9	Operational performance and reliability demonstrated for an array of type machines ( <b>3–5 devices</b> ).	Fully de-risked business plan for utility-scale deployment of arrays.

in the TRL-TPL space, together with potential measures to bridge the VoD for new ventures.

### 5.1 | Technology readiness level

TRL was initially developed by NASA to estimate the maturity of a technology of high risk, novelty and complexity, for example, for space programmes. For WEC technology, the technology readiness assessment is tailored by Fitzgerald and Bolund [198], in the categories of functional readiness and lifecycle readiness, as shown in Table 1. Functional readiness is more commonly used than lifecycle readiness, as there only exists very limited experience of long-term WEC operation. Hereafter, TRL refers to the functional readiness only. Table 1 clearly shows that TRL significantly relates to the scale ratio, mainly according to the Froude number, with successful demonstration of a small WEC array at TRL 9 indicating that the technology is mature, or ready for commercialisation.

A WEC development roadmap, from design to commercialisation, is also discussed in [199], with specific foci on TRLrelated development activities and assessment criteria for single WECs and WEC farms. By summarising existing R&D WEC projects, the WEC development plan (in 2010) is divided into five stages [200] and six stages in 2021 [66], aiming to comprise the best practices and recommended procedures for wave energy technology. One big lesson learnt for the TRL-roadmap in [200] is that the development cost and time are high, as shown in Figure 9(a).

#### 5.2 | Technology performance level

Since TRL was initially used for the NASA space program, development cost and time are not as important as technology maturity. However, this is not the case for WEC technology. Marketability and affordability are as important as technology maturity, which are not considered in the TRL assessment of Table 1. Thus, the concept of TPL was proposed by Weber [201, 202], to address the importance of technology performance, power capability, system availability, CapEx and OpEx. The characteristics and categories of each TPL level are detailed in Table 2. To assess TPL at early stages of WEC projects, guidance is given in [203].

With more emphasis on LCoE drivers, an updated TPL assessment method was used for the Wave Energy Prize [204]. For successful WEC commercialisation, stakeholder requirements on WEC technology are identified in [64], as: (i) having market-competitive cost of energy, (ii) providing a secure investment opportunity, (iii) being reliable for grid operations, (iv) benefiting society, (v) being acceptable to permitting and certification; (vi) being safe, and (vii) being deployable globally. A third TPL assessment update is provided in [205], where TPLs can be applied to all TRLs and development stages of WEC devices and farms. At low TRL, TPL assessment is very effective, as it considers a wide range of techno-economic performance criteria. At high TRLs, TPL assessment is more strict. More detailed methods to score TPL is given in [206].

#### 5.3 | Development trajectory

Based on TRL and TPL assessment, a TRL-TPL matrix is established in [65, 202], to discuss possible development trajectories for a WEC project. Intuitively, there are two simple development trajectories: the TRL-first trajectory (blue curve) and the TPL-first trajectory (green curve), as shown in Figure 9(b). The TRL-first trajectory is conventionally used in WEC development, where WEC development concentrates on improving the TRL first, and then attempts to improve TPL at a high TRL. In contrast, the TPL-first trajectory prioritises TPL over TRL by evaluating WEC techno-economic



**FIGURE 9** (a) WEC development cost and elapsed time evolution as TRL increases from 1 to 9, data from [200], and (b) potential WEC development trajectories in the TRL-TPL-matrix. In (a), the mean values of development cost and elapsed time are used. In (b), the development trajectory of the *SEAREV* device is adapted from [26]. The markers P1 and P2 show the development cost and elapsed time at TRL levels of 3 and 8, respectively

TPL	Characteristics	Category
1	Majority of key performance characteristics and cost drivers do not satisfy and present a barrier to potential economic viability.	<b>High</b> : Technology is economically viable and competitive as a renewable energy form.
2	Some of key performance characteristics and cost drivers do not satisfy potential economic viability.	
3	Minority of key performance characteristics and cost drivers do not satisfy potential economic viability.	
4	In order to achieve economical viability under distinctive and favourable market and operational conditions, some key technology implementation and fundamental conceptual improvements are required.	Medium: Technology features some characteristics for potential economic viability under distinctive market and operational conditions. Technological or conceptual improvements may be required.
5	In order to achieve economic viability under distinctive and favourable market and operational conditions, some key technology implementation improvements are required.	
6	Majority of key performance characteristics and cost drivers satisfy potential economic viability under distinctive and favourable market and operational conditions.	
7	Competitive with other renewable energy sources given favourable support mechanism.	Low: Technology is not economically viable.
8	Competitive with other energy sources given sustainable support mechanism.	
9	Competitive with other energy sources without special support mechanism.	

<b>TABLE 2</b> Characteristics and categorise of TPLs, adapted from	ı [202]
---	---------

performance at low TRLs, with significantly lower design cost implications. In addition, another combined trajectory is discussed also in [65], and demonstrated well by the *SEAREV* case study [26], shown as the black curve in Figure 9(b). The first generation device, *SEAREVG1*, reaches (TRL,TPL)=(4,2), but its LCoE is high. Shape optimisation and control are implemented for the second-generation device, *SEAREV G21*, at (TRL,TPL)=(3,3). Further design optimisation for the third-generation device, *SEAREV G3*, prioritises TPL over TRL, arriving at (TRL,TPL)=(2,6).

The traditional TRL-first trajectory requires several technology steps at full technology readiness. Development costs, and time required for each TRL, are estimated in [200], and the mean values are illustrated in Figure 9(a). As the TRL increases from 3 to 8 (see the points P1 and P2), the accumulated budget and time increase from 0.1 M€ to 25.8 M€ and from 0.5 years to 6.4 years, respectively. By projecting P1 and P2 to the TPL-first and TRL-first trajectories in Figure 9(b), respectively, it is clear that the TPL-first trajectory is significantly cheaper in achieving a high TPL [202]. A comparison study of the TRL-first, TPL-first, and combined trajectories in [65] strongly recommends the TPL-first trajectory, as the other requires double the development time and cost, and is more prone to project failure. Based on the TRL concept, a framework for evaluating WEC

**TABLE 3** WEC development stages based on TRLs, inspired by [66]

Stage	TRL	Verification
0: Concept creation	1	Analytical and numerical models
1: Concept development	2-3	
2: Design optimisation	4	Experimental tests in controlled environment
3: Scaled demonstration	5-6	
4: Commercial-scale single device demonstration	7—8	Experimental tests in representative environment
5: Commercial-scale array demonstration	9	

performance, and a development process, is identified in [66]. Within this framework, the WEC development process is divided into six TRL-related stages, given in Table 3. For each stage, technology performance is evaluated in terms of power capture, power conversion, controllability, reliability, survivability, maintainability, installation ability, manufacturability, and affordability, to define evaluation criteria, methods, thresholds, activities and stage entry requirements. R&D projects should proceed further to the next TRL-based stage only when all stage gate metrics are met.

In the TRL-TPL matrix, market entry conditions are defined as (TRL,TPL)=(9,>7) [202]. However, this does not indicate successful commercialisation. According to the commercial readiness index (CMI) defined in [207, 208], market-entry WEC technology only arrives at the commercial trial stage, far away from successful commercialisation.

### 5.4 | Valley of death

The VoD, for WEC commercialisation, is referred to as the period in which a startup venture begins to develop products for early commercial trial, but has not yet brought in revenue from market. As shown in Figure 10, the VoD refers to the negative cash flow gap between R&D activities to early commercialisation. In this study, cash flow (CF) [63] is defined as

$$CF(y) = \frac{Rev(y) - CapEx(y) - OpEx(y) - Tax(y)}{\left(1 + \frac{R_{\rm d}}{100}\right)^{y}},$$
 (30)

where *Rev*, *CapEx*, *OpEx* and *Tax* are the annual revenue, capital expenditure, operation and maintenance expenditure, and annual taxes, respectively.  $R_d$  and y are the discount rate and the year index, respectively, with the life-cycle running from  $y_0$  to Y.

In Figure 10, basic success is represented by the black curve, where a positive CF is achieved at the end of commercialisation. For some specific WEC projects or devices, failure may occur before operation or generation of any revenue (see the red curve), mainly due to a low TPL, for example, low in manufacturability, installability or grid accessibility. After successful commissioning, a WEC project starts to operate, generating elec-



FIGURE 10 The cash flow valley of death

tricity, and bringing revenue. Still, commercialisation may fail if the deployed WEC technology is of low survivability, shown by the brownish red curve. WEC devices may also be destroyed by extreme waves conditions, for example, storms. Another big concern is marketability. Even though a positive CF is achieved, project failure may still occur (see the grey curve) if the LCoE from wave is not as competitive as other renewable energy technologies. Such a failure may be caused by frequently required maintenance, disappearance of feed-in tariff (FIT), or a dramatic reduction in the LCoE of other renewable energy technologies.

To successfully commercialise wave energy technology, the IAG collaboration plays the major role to bridge the VoD. Technically, novel WEC concepts, PTO innovation and robust control show great potential in LCoE reduction. In addition, the TPL-first development trajectory can further reduce the LCoE by saving development cost and time, and mitigating development risk. With these technical measures, the black CF curve can be raised to the green curve, thus relieving the dearth of investment during the VoD. On the other hand, market incentives from government-related sectors, like FIT, can benefit revenue and CF directly, lifting the black curve to the blue one. FITs have a significant influence on overall WEC economic performance [63], and an appropriate FIT rate can even drive a defective commercial project into a profitable one. Beyond market incentives, government-related sectors can further develop regional and national strategies to reinforce IAG collaboration, to achieve a significant commercial success, shown as the dark green curve in Figure 10.

To bridge the VoD via IAG collaboration, public or government-related sectors play a most important role to bring together researchers and investors, by developing national strategies and using market incentives, even at the early stage of a specific WEC technology for better two-way knowledge transfer and communication [8, 209]. Current national strategies and market incentives are discussed in Section 8. Given that the development and commercialisation of WEC technology require significant time and funding levels, and are of high risk, public investment should be prolonged even after market entry, to mitigate investment risk by building sea test sites,



FIGURE 11 Historical development of wave energy technology

establishing logistics chains, providing grid connection and easing legal permitting. With better understanding of WEC technology and its mitigated risk, private investment, that is, seed investment, venture capital and stock investment, may get involved at early stages of WEC technology development. Thus, such a reinforced IAG collaboration shows a possibility to bridge the VoD for commercial success.

### 6 | HISTORICAL AND COMMERCIAL EFFORT

This section summarises the historical and ongoing development of wave energy technology, to address both academic and commercial milestones.

# 6.1 | Historical development of wave energy technology

The idea to transfer wave energy into a useful form is not new, dating back to 1799. Since then, the historical development of wave energy technology is divided into six eras, shown in Figure 11. A notable overview of WEC history development is summarised in [138]. The years of 1973, 1985, 1998, 2012 and 2016 are treated as turning points for WEC development, also used in Figure 11. R&D and commercial activities in each era are detailed in the following subsections.

#### 6.1.1 | 'Pre-history' Era, before 1973

In this era, R&D and commercial activities are not well documented, and most research work is typified by trial-and-error methods. However, there are still some significant fundamental achievements, including: (i) the first patent published in France in 1799 [210]; (ii) the first practical *wave motor* device in the United States, operating from 1898 to 1910 [138]; (iii) the OWCbased navigation buoy developed in Japan from 1945 to 1965, is successfully commercialised with more than 1000 devices deployed worldwide [138]. To the best of the authors' knowledge, the OWC-based navigation buoy is the only successfully commercial WEC device to date.

#### 6.1.2 | Modern Era, 1973–1985

With oil price rising sharply from 21 to 57 \$ per barrel in 1973, several countries invest heavily in renewable energy technologies. The landmark of WEC technology entering into the

3079

TABLE 4	Some typical	pre-commercial	wave energy	converters
IADLE 4	Some typical	pre-commercial	wave energy	converters

	-		_		Capacity			
WEC	Country	Developer	Туре	Stage	(kW)	Year	Status	Reference
KAIMEI	Japan	JAMSTEC	OWC	4	40	1978-1979	decommissioned	[35, 222]
TAPCHAN	Norway	Norway A.S.	TWEC	4	385	1985-1989	destroyed by storm	[218, 223]
Kvaerner OWC	Norway	Kvaerner Brug A/S.	OWC	4	500	1985-1989	destroyed by storm	[138, 219]
Islay OWC	UK	QUB, Wavegen	OWC	4	75	1991-2000	replaced by LIMPET	[224]
OSPREY	UK	Wavegen	OWC	4	2000	1995	lost during installation	[32, 138]
Mighty Whale	Japan	JAMSTEC	OWC	5	110	1998-2000	decommissioned	[225, 226]
Pico OWC	Portugal	IST, WavEC	OWC	4	400	1999; 20016-2018	turbine fault;	
damaged by storm	[220]							
LIMPET	UK	QUB, Wavegen	OWC	4	500	2000-2012	stopped	[224, 227]
AWS	Netherlands	Teamwork Technology	PA	4	1000	2001;2002; 2004	lost due to	
pump failure	[228, 229]							
Pelamis	UK	Pelamis Wave Power	AWEC	5	750	2004-2007; 2010-2011	structure failure 2011	[230]
SeaBased	Sweden	Seabased Industry	PA	5	$10 \times 10$	2005-2007	decommissioned	[231, 232]
WaveRoller	Finland	AW-Energy OY	OWSC	4	350	2007-2008	decommissioned	[233]
PowerBuoy	United States	OPT	PA	4	40	2009-2010	decommissioned	[234]
Mutriku	Spain	EVE	OWC	5	296	2009-	active	[235, 236]
Oyster 800	UK	Aquamarine Power	OWSC	4	800	2012-2015	stopped	[237]
BOLT LIfeSaver	Norway	Fred. Olsen	PA	4	30	2015-2016	stopped	[238, 239]
green WAVE	Australia	Oceanlinx	OWC	4	1000	2014	damaged during transportation	[240]
Eagle Wanshan	China	GIEC	AWEC	4	100	2015-2016	stopped	[241]

modern era, is the proposal of *Salter's Duck*, whose hydrodynamic efficiency is tested up to 80% [211]. In this era, theoretic fundamentals are established, including: (i) the concepts of 'resonance', 'absorption length' and 'power optimisation' are defined for the first time in [168]; (ii) the theoretical maxima of absorption are derived in [212]; (iii) latching control [213] and reactive control [214] are tested; (iv) the constructive park effect of WEC arrays is studied for the first time in [215]; and (v) the first economic study of WEC technology is given in [216]. In addition, there is also some practical progress, represented by the sea trial of the *KAIMEI OWC* and *Cockerell Raft* concepts.

#### 6.1.3 | Trough Era, 1985–1998

In 1985, the oil price drops from 71 to 25 \$ per barrel, and activity in renewable energy decreases dramatically. However, there are still some noteworthy milestones. In theory, feed-forward control is studied to overcome the non-causality [217], while several devices are tested in the open ocean, including the *TAPCHAN* [218], *Kvaerner column* [219], *OSPREY OWC* [32], *Islay OWC* [32] and *Pico OWC* [220].

#### 6.1.4 | Explosion Era, 1998–2012

The Kyoto Protocol is signed in 1998 and carbon emission reduction becomes an international imperative, marking 1998 as

the start of the explosion era of WEC technology. A significant milestone is the development of the *Pelamis* device, considered as the most promising WEC technology for commercial application. In this era, pre-commercial activities are stimulated by regional and national support programmes and market incentives, characterised by: (i) WEC companies developing some well-known pre-commercial devices, further detailed in Table 4; (ii) WEC technology evolves from onshore to offshore, from small to large capacity; (iii) a number of open sea testing sites are commissioned with grid connections, with EMEC opening in 2004 as probably the first and most developed test site; and (iv) a number of structure failures are reported, with high financial loss, or total device loss, leading to adverse publicity.

#### 6.1.5 | Distrust Era, 2012–2016

In 2012, *Statkraft* terminated its ocean energy programme, tipping the first domino and opening the distrust era. Several companies, even some highly rated ones, fail to pass through the VoD and go into bankrupcy or liquidation, for example, *Wave-bob Ltd., AWS, Wavegen, Pelamis Wave Power, Oceanlinx,* and *Aquamarine Power.* This bad news reduces public and private investor trust, given that the total investment in these companies was significant, for example, about 64 M€ for *Oceanlinx,* and about 81 M€ for *Pelamis.* One positive trend in this era is that some companies turned to niche markets, for example, the *OPT PB3* and *PH4S* buoys for powering ocean observation devices.



FIGURE 12 Photos of some pre-commercial WEC devices in Table 4, including the KAIMEI [36], LIMPET [36], Mutriku [36], TAPCHAN [242], BOLT LIfeSaver [242], AWS [242], SeaBased [243], PowerBuoy [244], Pelamis [230], and Oyster [242] devices

#### 6.1.6 | Reboot Era, 2016–present

In 2016, the Paris Agreement entered into force, producing consequent activity increases in regional and national support schemes. Meanwhile, *WaveStar* becomes inactive after 13 years of operation, following 40 M€ of investment. Then, *Carnegie Clean Energy* receives more than 39 M€ total investment, and wave energy was successfully applied to an aquaculture platform. However, *Seabased* closes its production facility in Sweden and the *Pico* plant suffers from structural damage in a storm. Following the Paris Agreement, more than 110 countries pledge to reach zero carbon emission by 2050, suggesting a boom in public and private renewable energy investment. Even though wave energy is more challenging to harvest than other renewable energy resources, WEC technology is well poised to be rebooted by increased national support strategies and market incentives, and reinforced IAG collaboration.

# 6.2 | Pre-commercial development of wave energy

As mentioned in Section 6.1, the OWC navigation buoy developed in Japan is the only successful commercialisation case, with no full-scale commercial WEC farm in operation. *Commercial devices* refer to WECs that are: (i) characterised by high TRL and TPLs, that is, (TRL,TPL)=(9,>7); (ii) fully functioning, with affordable electricity for utility markets or reliable power supply for niche markets; and (iii) market accessible with specific product availability. Current WECs cannot fully satisfy (ii) and (iii), though some can meet (i), referred as pre-commercial devices.

This subsection only summarises pre-commercial WEC projects and related R&D activities, with some well-known precommercial WECs listed in Table 4 and Figure 12. In Table 4, the development stage is defined according to the TRL [66], as shown in Table 3. It can been seen that only the *Mighty Whale*, *Pelamis, Mutriku* and *SeaBased* devices are at stage 5. Most precommercial devices are based on OWC or PA concepts, showing high consistency to the R&D foci reviewed in [11, 62, 221].

#### 6.3 | Prospects for commercialisation

Learning from the operational experience of pre-commercial WECs in Table 4, some recommendations for commercialisation may be made: (i) System survivability should be the most important concern for commercialisation, as several WECs suffered from structure failure in storms. (ii) The WECs in Table 4 require significant development funding and time, mostly developed along the TRL-first trajectory. Without considering TPL at early stage of WEC development, WEC projects may fail purely due to poor installability or transportability, for example, the OSPREY and green WAVE devices. Thus, the TPL-first development trajectory is strongly recommended to save development cost and time, and to mitigate development risk. (iii) The real performance of various pre-commercial WECs is not as optimistic as expected, potentially due to optimistic power production estimates (from linear models) or capacity factors. The capacity factor in long-term testing is low [220, 236], for example, 0.11 [236], but is over-optimistically estimated as 0.3 in LCoE assessments [59, 245]. Efforts to reduce the uncertainty in LCoE assessment are important in improving investment decisions and investor confidence.

# 7 | MARKET OPPORTUNITY FOR WAVE ENERGY

The dominant target market for wave energy technology is utility-scale electricity, though R&D and commercialisation activities towards niche markets are also emerging. Both of these market opportunities are now separately considered.

### 7.1 | Utility market

As mentioned in Section 1, there exists a conflict between the increasing global energy demand and carbon reduction promises [3], with a focus on renewable energy technologies to provide carbon-free electricity. However, the current LCoE of wave energy is estimated at a high level, ranging 120–470



FIGURE 13 LCoEs of different technologies in the United States, with or without carbon pricing, in black and red, respectively, data from [246]

\$/MWh (about 100–400€/MWh) [59]. Compared with other mature renewable energy resources, for example, solar and wind power, or fossil fuels, wave power is not competitive or market viable in the utility market [246], as shown in Figure 13. If carbon pricing is applied to the energy technologies in Figure 13, renewable energy resources will have lower an LCoE than fossil-based resources. In this scenario, wave energy technology is, indeed, marketable.

As wave energy technology is untapped, its LCoE can be further reduced to 100–300 \$/MWh (about 84–252 €/MWh) for GWs of installed capacity, and to 100–150 \$/MWh (about 84–126 €/MWh) for 10 GW installed capacity [59]. With accumulated operation experience, a recent OES annual report projected that the LCOE of wave energy can be reduced to 100– 150€/MWh by 2030–2035 [247]. Thus, electricity from wave energy is expected to be competitive in the utility market.

In addition, wave energy can be treated as a complementary source for offshore wind farms [14, 17], and the combination of wind and wave energy results in legislative and technical synergies for both technologies [15], to reduce the LCoE further. However, such a combination strongly depends on installation sites [18], and an ideal site, characterised by less extremes, elevated mean values, stable behaviour and low correlation, will result in a more smooth output and fewer hours of zero production. The Irish coast is such an ideal site for wind-wave integration [14, 17], where wind and wave resources are low correlated, and joint wind-wave farms can mitigate against the highfrequency variability in both resources.

WEC technology, with accumulated install capacity at GWs, shows a great potential for the utility market by providing carbon-free and affordable electric, but still unattractive to private investors, as its R&D and commercialisation activities are still costly and risky to invest in.

### 7.2 | Niche market

As mentioned in Section 6.1, the only successful commercialisation of WEC technology is the OWC navigation buoy, which belongs to a niche market rather than the utility market. For the past decade, many countries have established regional or national strategies to develop their 'blue economy'. Thus, oceanbased applications, for example, ocean observation and desalination, and fish farming, are growing, and require an economical and clean power supply [248]. Wave energy can meet such energy demands to advance the blue economy, where alternative (especially conventional) energy sources can be prohibitively expensive, due to the relatively remote consumption point.

Ocean-related niche markets include: (i) ocean navigation and observation [249], (ii) coastal protection [250–252], (iii) desalination [253, 254], (iv) island micro-grid [138], and (v) marine aquaculture, (vi) multi-function offshore platforms [255, 256], and (vii) other applications, for example, underwater vehicle charging, disaster recovery and resiliency, seawater mining and marine algae [248]. These potential niche markets are well summarised in [22, 221, 257].

Compared with the utility market, the rated capacity for niche markets is much smaller, ranging from several Watts to hundreds of kW. Such a relatively small capacity may result in a small geometric dimension, consequently reducing development time, cost and risk, which make it more appealing to public and private investment, with potential investors already coming from financially secure application domains, showing strong potential to pass through the VoD to a commercial success. It is also expected that rapid growth in wave energy niche market applications can assist the development effort for the utility market, by accumulating operation and maintenance experience, as well as WEC system design expertise.

#### 7.3 | Prospects for commercialisation

To sum up, the potential size of the utility market is up to TWs, but current wave energy technology has not fully demonstrated its competitiveness with respect to other energy technologies. As technical, economical and administrative challenges coexist, reinforced IAG collaboration is strongly recommended to advance the TPL of wave farms for the utility market. Niche markets in ocean applications have been emerging, and wave energy shows promising potential in providing clean and economic power supply for ocean-based applications. However, the market size is still unknown, and only limited operational experience is available, creating cost uncertainties. Longer term operations are required to further quantify the commercial potential of wave energy technology for niche markets.

## 8 | FACTORS AFFECTING THE DEVELOPMENT OF WAVE ENERGY

Recalling the historical development of wave energy technology, as documented in Figure 11, key factors that significantly influence the development of wave energy technology can be separated into external factors, for example, fossil fuel price, development of other renewable energy technologies, and national/community factors, for example, supporting strategies and market incentives, which are discussed in the following two subsections.

TABLE 5 National schemes for ocean energy, data from [260]

	National strategy				Market incentive				
Country	Capacity Target	National Action Plan	Technology Roadmap	Marine Spatial Plan	Feed-in Tariff	Contract for Dif- ference	Green Certifi- cate	Quota obliga- tion	Renewable Energy Auction
Belgium			•	•			•		
Canada	•	•	•	•	•				
China	•	•		•			•		
Denmark			•						
France	•			UD	•				
Germany				•	•				
Ireland	•	•	•	UD	•				
Italy	•				•				
Japan		•	•		•				
Korea	•	•	•					•	
Mexico	•		•				•		
Netherlands				•	•				
Monaco				•					
Norway				•			•		
New Zealand				•					
Portugal	•	•	•	•					
South Africa				•					
Spain	•								•
Sweden		•		UD			•		
UK	•	•	•	•		•			
United States		•		•					

#### 8.1 | External factors

In Figure 11, 1973 and 1985 can be clearly identified as turning points, mainly due to the rapid changes in oil price in those years. Oil prices surged from 21 to 57 \$ per barrel in 1973, resulting in a corresponding surge in wave energy development, while the price collapsed from 71 to 25 \$ per barrel in 1985 consequently disincentivised R&D development. Although fossil fuel prices comprise one of the most important factors, such a causal factor is somewhat unpredictable, though (despite new recovery methods such as fracking) one can only imagine that fossil fuels will become increasingly more expensive, with dwindling supply.

In Figure 11, 2012 is also marked as a tipping point, in which a series of WEC company failures were indicative of the currently low marketability of WEC technology in the utility market. Compared with other renewable energy technologies, for example, wind and solar power, current WEC technology is untapped and uncompetitive with a high LCoE, as shown in Figure 13. One hard lesson learnt from some failed projects is that the TPL-first development trajectory should be used, to address technology performance at early development stages. With costs scaling up exponentially with scale, a premature rush to high TRL levels has been shown to be costly. The repercussions of some high profile WEC company failures are still felt in the wave energy community. Perhaps, over-optimistically, wave energy technology is projected to achieve a competitive LCoE at 100-150 €/MWh by 2030–2035 [247].

A further external factor is the rapid rise in other renewable energy technologies, for example, in offshore wind (including floating offshore wind) [258, 259], which builds on many years of expertise experience in wind energy, with incremental technical problems only to be solved, while wave energy still wrestles with fundamental issues. The rapid acceleration in offshore wind has garnered both offshore technologists and investors from the wave energy sector.

#### 8.2 | National strategies and market incentive

For public investors, for example, governments, the benefits of investing in wave energy are many, including (i) environmental benefit, achieving carbon neutrality; (ii) broadening of the energy mix and provision of greater energy security; and (iii) economic benefit, for example, industry and job creation,

**TABLE 6**Feed-in tariff in some countries, data from [260, 264 268, 269]

Country	Strategy	Capacity (MW)	rate (€/MWh)	duration (Year)
Canada	Ontario FIT	<5.5	170	40
China	FIT	-	330	-
France	FIT	-	173	_
Germany	FIT	<50;>50	124; 34.7	-
Ireland	FIT	-	260	_
Italy	FIT	< 5; >5	300; 190	15
20				
Netherlands	Subsidy	-	130	_
Philippines	FIT	-	310	-
UK	CfD	_	360	-

blue economy. Based on the OES annual report in 2017, some national supporting schemes for ocean energy are detailed in Table 5, within which UD means under development.

In Table 5, national strategies include: (i) capacity targets, to express national commitment to ocean energy deployment; (ii) action plans agreed by public and private sectors to facilitate deployment; (iii) roadmaps, providing long-term frameworks for developing policies and supporting actions; and (iv) marine spatial plans, to remove administrative barriers. As ocean energy resources and market data vary from country to country, national strategies developed by each country also vary and are detailed in [260]. Roadmaps with specified long-term pathways are important to mobilise national efforts to improve ocean technology, which are articulated in a number of countries, for example, UK [261], Denmark [262], and Ireland [263]. Additionally, national policies for innovation, manufacturing and deployment of ocean energy are discussed in [264].

Table 5 also summarises several commonly applied market incentives, of which the FIT is the most common supporting measure, as it can directly improve the profitability of ocean energy projects. Since 2014, the UK has provided the 'contract for difference (CfD)' mechanism to replace its original used FIT scheme [260]. The FIT and CfD schemes belong to market-pull incentive, while the others fall into the market-push class [265]. FIT is one of the most successful incentive schemes for promoting the growth in wind and solar power [266], and is naturally expected to have significant influence on encouraging private investment in wave energy. For wave energy technology, sensitivity of net present value to FIT variation is analysed in [63, 267], which also address the effectiveness of FIT schemes, in relation to its impact sensitivity on project profitability [63]. This reveals some of the rationale for policy makers to applying FIT schemes to encourage private investment. Some active FIT supporting schemes are summarised in Table 6, which can prolong public investment and encourage corresponding private investment to bridge the VoD.

Occasionally, governments invest directly in interventions and mechanisms that accelerate the development of wave energy intellectual property (IP), in addition to providing FIT GUO AND RINGWOOD

support schemes. Such interventions are somewhat altruistic, since the benefits of directly supporting fundamental technology development can be potentially enjoyed by many other jurisdictions. However, maintaining and supporting wave energy IP development directly brings the capability to generate a significant export industry and supply chain, which is of potentially greater long-term value than the development of wave farms locally. Although many jurisdictions provide general funding schemes for both R&D and commercialisation, Wave Energy Scotland is somewhat unique in dedicating funding to wave energy IP development, originally founded to retain and manage the IP held by Scottish companies Pelamis Wave Power and Aquamarine Power, following their demise in 2014 and 2015, respectively. In a more measured way, some jurisdictions have included ocean energy as one of the national research priorities, for example by Science Foundation Ireland in 2013, giving some level of preferential treatment to research and R&D proposals in this area.

### 9 | CONCLUSIONS

This review summaries the historical and ongoing research and commercialisation efforts devoted to wave energy technology. Significant spatial and temporal variability in the wave power resource has a fundamental role in diversifying the development of successful WEC concepts, with a need for a collective approach to common fundamental issues, such as modelling, PTO and control design, survivability, and performance metrics. Regarding technology development trajectories, TPL must clearly be prioritised over TRL, particularly at early project stages. Clearly, investor risk must be reduced by providing more certainty in national and international support programmes, focussing on common technological challenges, to reduce LCOE, LCOE uncertainty and to examine limitations in supply chains and marine licensing arrangements, and maximisation of the potential of IAG collaboration.

Historical analysis has shown that survivability and installability are key metrics, which not only affect the economics and success of individual wave energy projects, but also play a large role in sector confidence and investability. With increasing emphasis on the provision of carbon-free energy, and a need to diversify the mix of renewable energy sources, wave energy is well poised to supplement, and complement, existing and more mature renewable energy technologies. The next decade will be crucial in deciding if wave energy can make the breakthrough needed to become a mainstream renewable energy technology.

### ACKNOWLEDGEMENTS

This document is the result of a research project funded by the Marie Skłodowska-Curie Action (Grant No. 841388) and Science Foundation Ireland (Grant No. SFI/13/IA/1886, and Grant No. 12/RC/2302-P2 for the Marine and Renewable Energy Ireland Centre).

#### **CONFLICT OF INTEREST**

The authors have declared no conflict of interest.

#### DATA AVAILABILITY STATEMENT

Data sharing not applicable as no new data generated, or the article describes entirely theoretical research.

#### ORCID

Bingyong Guo b https://orcid.org/0000-0003-3134-0043 John V. Ringwood b https://orcid.org/0000-0003-0395-7943

#### REFERENCES

- Guterres, A.: Carbon Neutrality by 2050: the World's most urgent mission. (United Nations Secretary General, 2020.) https://www.un.org/sg/en/content/sg/articles/2020-12-11/ carbon-neutrality-2050-theworld%E2%80%99s-most-urgent-mission (2020). Accessed 10 Feb 2021
- IEA: World Energy Outlook 2019. International Energy Agency, Paris (2019)
- IRENA: Global Renewables Outlook: Energy Transformation 2050 (2020)
- Reguero, B.G., Losada, I.J., Méndez, F.J.: A global wave power resource and its seasonal, interannual and long-term variability. Appl. Energy 148, 366–380 (2015)
- Mork, G., Barstow, S., Kabuth, A. & Pontes, T.: Assessing the global wave energy potential. In: International Conference on Ocean, Offshore and Arctic Engineering, pp. 447–454. Shanghai, China (2010)
- Gunn, K., Stock Williams, C.: Quantifying the global wave power resource. Renew. Energy 44, 296–304 (2012)
- 7. OES: An International Vision for Ocean Energy (2017)
- 8. IRENA: Innovation Outlook: Ocean Energy Technologies (2020)
- IEA: Electricity Information Overview 2020 Edition. International Energy Agency. https://webstore.iea.org/ electricity-information-overview-2020-edition (2020). Accessed 10 Feb 2021
- McCormick, M.E.: Ocean Wave Energy Conversion. Wiley, New York (1981)
- López, I., Andreu, J., Ceballos, S., De Alegría, I.M., Kortabarria, I.: Review of wave energy technologies and the necessary power-equipment. Renew. Sustain. Energy Rev. 27, 413–434 (2013)
- Langhamer, O., Haikonen, K., Sundberg, J.: Wave power–Sustainable energy or environmentally costly? A review with special emphasis on linear wave energy converters. Renew. Sustain. Energy Rev. 14(4), 1329– 1335 (2010)
- Copping, A.E., Hemery, L.G., Overhus, D.M., Garavelli, L., Freeman, M.C., Whiting, J.M., et al.: Potential environmental effects of marine renewable energy development—the state of the science. J. Mar. Sci. Eng. 8(11), 879 (2020)
- Fusco, F., Nolan, G., Ringwood, J.V.: Variability reduction through optimal combination of wind/wave resources–An Irish case study. Energy 35(1), 314–325 (2010)
- Pérez Collazo, C., Greaves, D., Iglesias, G.: A review of combined wave and offshore wind energy. Renew. Sustain. Energy Rev. 42, 141–153 (2015)
- Widén, J., Carpman, N., Castellucci, V., Lingfors, D., Olauson, J., Remouit, F., et al.: Variability assessment and forecasting of renewables: a review for solar, wind, wave and tidal resources. Renew. Sustain. Energy Rev. 44, 356–375 (2015)
- Gallagher, S., Tiron, R., Whelan, E., Gleeson, E., Dias, F., McGrath, R.: The nearshore wind and wave energy potential of Ireland: a high resolution assessment of availability and accessibility. Renew. Energy 88, 494– 516 (2016)
- Kalogeri, C., Galanis, G., Spyrou, C., Diamantis, D., Baladima, F., Koukoula, M., et al.: Assessing the European offshore wind and wave energy resource for combined exploitation. Renew. Energy 101, 244–264 (2017)
- Weiss, C., Guanche, R., Ondiviela, B., Castellanos, O.F., Juanes, J.: Marine renewable energy potential: a global perspective for offshore

wind and wave exploitation. Energy Convers. Manage. 177, 43-54 (2018)

- Fernandez Chozas, J., Soerensen, H.C., Kofoed, J.: Predictability and variability of wave and wind: wave and wind forecasting and diversified energy systems in the Danish North Sea. Technical Report (2013)
- Sasaki, W: Predictability of global offshore wind and wave power. Int. J. Mar. Energy 17, 98–109 (2017)
- IRENA: Fostering a blue economy: offshore renewable energy. International Renewable Energy Agency. https://www.irena.org/publications/2020/Dec/Fostering-a-blue-economy-Offshore-renewable-energy (2020). Accessed 10 Feb 2021.
- Koca, K., Kortenhaus, A., Oumeraci, H., Zanuttigh, B., Angelelli, E. & Cantu, M. et al.: Recent advances in the development of wave energy converters. In: European Wave and Tidal Energy Conference, pp. 1–10. Aalborg, Denmark (2013)
- Stansby, P., Moreno, E.C., Stallard, T.: Large capacity multi-float configurations for the wave energy converter M4 using a time-domain linear diffraction model. Appl. Ocean Res. 68, 53–64 (2017)
- Moreno, E.C., Stansby, P.: The 6-float wave energy converter m4: ocean basin tests giving capture width, response and energy yield for several sites. Renew. Sustain. Energy Rev. 104, 307–318 (2019)
- Cordonnier, J., Gorintin, F., De Cagny, A., Clément, A.H., Babarit, A.: SEAREV: case study of the development of a wave energy converter. Renew. Energy 80, 40–52 (2015)
- Giorgi, G., Penalba, M. & Ringwood, J.V.: Nonlinear hydrodynamic models for heaving buoy wave energy converters. In: Asian Wave and Tidal Energy Conference, pp. 1–10. Marina Bay Sands, Singapore (2016)
- Davidson, J., Windt, C., Giorgi, G., Genest, R. & Ringwood, J.V.: Evaluation of energy maximising control systems for wave energy converters using OpenFOAM. In: Nóbrega, J., Jasak, H. (eds.) OpenFOAM Workshop, pp. 157–171. Springer, Cham (2019)
- Smith, H.C., Haverson, D., Smith, G.H.: A wave energy resource assessment case study: review, analysis and lessons learnt. Renew. Energy 60, 510–521 (2013)
- Lavidas, G., Venugopal, V.: Application of numerical wave models at European coastlines: a review. Renew. Sustain. Energy Rev. 92, 489–500 (2018)
- Guillou, N., Lavidas, G., Chapalain, G.: Wave energy resource assessment for exploitation-a review. J. Mar. Sci. Eng. 8(9), 705 (2020)
- Thorpe, T.: A brief review of wave energy. A report produced for UK Department of Trade and Industry (1999)
- Falnes, J.: A review of wave-energy extraction. Mar. Struct. 20(4), 185–201 (2007)
- Drew, B., Plummer, A., Sahinkaya, M.: A review of wave energy converter technology. P. I. Mech. Eng. A-J. Pow. 223(8), 887–902 (2009)
- Falcão, A.: Wave energy utilization: a review of the technologies. Renew. Sustain. Energy Rev. 14(3), 899–918 (2010)
- Falcão, A., Henriques, J.: Oscillating-water-column wave energy converters and air turbines: a review. Renew. Energy 85, 1391–1424 (2016)
- Portillo, J., Reis, P., Henriques, J., Gato, L., Falcão, A.: Backward bent-duct buoy or frontward bent-duct buoy? Review, assessment and optimisation. Renew. Sustain. Energy Rev. 112, 353–368 (2019)
- Sheng, W.: Wave energy conversion and hydrodynamics modelling technologies: a review. Renew. Sustain. Energy Rev. 109, 482–498 (2019)
- Babarit, A., Hals, J., Muliawan, M.J., Kurniawan, A., Moan, T., Krokstad, J.: Numerical benchmarking study of a selection of wave energy converters. Renew. Energy 41, 44–63 (2012)
- Folley, M., Babarit, A., Child, B., Forehand, D., Boyle, L. & Silverthorne, K. et al.: A review of numerical modelling of wave energy converter arrays. In: International Conference on Ocean, Offshore and Artic Engineering, pp. 535–545. ASME, Rio de Janeiro, Brazil (2012)
- Li, Y., Yu, Y.H.: A synthesis of numerical methods for modeling wave energy converter-point absorbers. Renew. Sust. Energy Rev. 16(6), 4352– 4364 (2012)
- Penalba Retes, M., Giorgi, G. & Ringwood, J.V.: A review of non-linear approaches for wave energy converter modelling. In: European Wave and Tidal Energy Conference, pp. 1–10. Nantes, France (2015)

- Penalba, M., Ringwood, J.V.: A review of wave-to-wire models for wave energy converters. Energies 9(7), 506 (2016)
- Folley, M.: Numerical Modelling of Wave Energy Converters: State-ofthe-Art Techniques for Single Devices and Arrays. Academic Press, Cambridge, MA (2016)
- Penalba, M., Giorgi, G., Ringwood, J.V.: Mathematical modelling of wave energy converters: A review of nonlinear approaches. Renew. Sustain. Energy Rev. 78, 1188–1207 (2017)
- Windt, C., Davidson, J., Ringwood, J.V.: High-fidelity numerical modelling of ocean wave energy systems: a review of computational fluid dynamicsbased numerical wave tanks. Renew. Sustain. Energy Rev. 93, 610–630 (2018)
- Papillon, L., Costello, R., Ringwood, J.V.: Boundary element and integral methods in potential flow theory: a review with a focus on wave energy applications. J. Ocean Eng. Mar. Energy 6, 303–337 (2020)
- Bard, J., Kracht, P.: Linear generator systems for wave energy converters. Dissertation, Louisiana State University (2013)
- Lin, Y., Bao, J., Liu, H., Li, W., Tu, L., Zhang, D.: Review of hydraulic transmission technologies for wave power generation. Renew. Sustain. Energy Rev. 50, 194–203 (2015)
- Ahamed, R., McKee, K., Howard, I.: Advancements of wave energy converters based on power take off (PTO) systems: a review. Ocean Eng. 204, 107248 (2020)
- Ringwood, J.V., Bacelli, G., Fusco, F.: Energy-maximizing control of wave-energy converters: the development of control system technology to optimize their operation. IEEE Control Syst. Mag. 34(5), 30–55 (2014)
- Bacelli, G., Ringwood, J.V.: Numerical optimal control of wave energy converters. IEEE Trans. Sustain. Energy 6(2), 294–302 (2014)
- Faedo, N., Olaya, S., Ringwood, J.V.: Optimal control, MPC and MPC-like algorithms for wave energy systems: an overview. IFAC J. Syst. Control 1, 37–56 (2017)
- Wang, L., Isberg, J., Tedeschi, E.: Review of control strategies for wave energy conversion systems and their validation: the wave-to-wire approach. Renew. Sustain. Energy Rev. 81, 366–379 (2018)
- Ringwood, J.V.: Wave energy control: status and perspectives 2020. IFAC-PapersOnLine 53, 1–12 (2020)
- Madhi, F., Yeung, R.W.: On survivability of asymmetric wave-energy converters in extreme waves. Renew. Energy 119, 891–909 (2018)
- Babarit, A.: A database of capture width ratio of wave energy converters. Renew. Energy 80, 610–628 (2015)
- Astariz, S., Iglesias, G.: The economics of wave energy: a review. Renew. Sustain. Energy Rev. 45, 397–408 (2015)
- OES: International Levelised Cost of Energy for Ocean Energy Technologies (2015)
- Davidson, J., Ringwood, J.V.: Mathematical modelling of mooring systems for wave energy converters–a review. Energies 10(5), 666 (2017)
- Garcia Teruel, A., Forehand, D.: A review of geometry optimisation of wave energy converters. Renew. Sustain. Energy Rev. 139, 110593 (2021)
- Guo, B., Ringwood, J.: Geometric optimisation of wave energy conversion devices: a survey. Appl. Energy 297, 117100 (2021)
- Teillant, B., Costello, R., Weber, J., Ringwood, J.: Productivity and economic assessment of wave energy projects through operational simulations. Renew. Energy 48, 220–230 (2012)
- Babarit, A., Bull, D., Dykes, K., Malins, R., Nielsen, K., Costello, R., et al.: Stakeholder requirements for commercially successful wave energy converter farms. Renew. Energy 113, 742–755 (2017)
- 65. Weber, J.W., Laird, D., Costello, R., Roberts, J., Bull, D. & Babarit, A. et al.: Cost, time, and risk assessment of different wave energy converter technology development trajectories. In: European Wave and Tidal Energy Conference, pp. 1–8. Cork, Ireland (2017)
- 66. OES: An International Evaluation and Guidance Framework for Ocean Energy Technology (2021)
- Pecher, A., Peter Kofoed, J.: Handbook of Ocean Wave Energy. Springer Nature, Basingstoke (2017)
- Falnes, J.: Ocean Waves and Oscillating Systems: Linear Interactions Including Wave-Energy Extraction. Cambridge University Press, Cambridge (2002)

- Stansberg, C., Contento, G., Hong, S., Irani, M., Ishida, S. & Mercier, R. et al.: The specialist committee on waves final report and recommendations to the 23rd ITTC. In: International Towing Tank Conference, pp. 505– 551. Venice, Italy (2002)
- Pierson, W.J., Moskowitz, L.: A proposed spectral form for fully developed wind seas based on the similarity theory of SA Kitaigorodskii. J. Geophys. Res. 69(24), 5181–5190 (1964)
- Hasselmann, K., Barnett, T., Bouws, E., Carlson, H., Cartwright, D., Enke, K., et al.: Measurements of wind-wave growth and swell decay during the Joint North Sea Wave Project (JONSWAP). Ergänzungsheft zur Deut. Hydrogr. Z. Reihe A 8(12), 1–95 (1973)
- Monárdez, P., Acuna, H., Scott, D.: Evaluation of the potential of wave energy in Chile. Proc. Int. Conf. Offshore Mech. Arct. Eng. 48234, 801– 809, (2008)
- Atan, R., Goggins, J., Nash, S.: A detailed assessment of the wave energy resource at the Atlantic Marine Energy Test Site. Energies 9(11), 1–29 (2016)
- Rusu, L., Onea, F.: The performance of some state-of-the-art wave energy converters in locations with the worldwide highest wave power. Renew. Sustain. Energy Rev. 75, 1348–1362 (2017)
- Bozzi, S., Besio, G., Passoni, G.: Wave power technologies for the Mediterranean offshore: scaling and performance analysis. Coast Eng. 136, 130–146 (2018)
- Fusco, F., Ringwood, J.V.: A study of the prediction requirements in realtime control of wave energy converters. IEEE Trans. Sustain. Energy 3(1), 176–184 (2011)
- Fusco, F., Ringwood, J.V.: Short-term wave forecasting for real-time control of wave energy converters. IEEE Trans. Sustain. Energy 1(2), 99–106 (2010)
- Nguyen, H.N., Tona, P.: Short-term wave force prediction for wave energy converter control. Control Eng. Pract. 75, 26–37 (2018)
- Pena Sanchez, Y., Garcia Abril, M., Paparella, F., Ringwood, J.V.: Estimation and forecasting of excitation force for arrays of wave energy devices. IEEE Trans. Sustain. Energy 9(4), 1672–1680 (2018)
- Schoen, M.P., Hals, J., Moan, T.: Wave prediction and robust control of heaving wave energy devices for irregular waves. EEE Trans. Energy Convers. 26(2), 627–638 (2011)
- Guo, B., Patton, R.J., Jin, S., Lan, J.: Numerical and experimental studies of excitation force approximation for wave energy conversion. Renew. Energy 125, 877–889 (2018)
- Shi, S., Patton, R.J., Abdelrahman, M. & Liu, Y.: Learning a predictionless resonating controller for wave energy converters. In: International Conference on Ocean, Offshore and Arctic Engineering, pp. 1–8. ASME, Glasgow, UK (2019)
- Mérigaud, A., Ramos, V., Paparella, F., Ringwood, J.V.: Ocean forecasting for wave energy production. J. Mar. Res. 75(3), 459–505 (2017)
- Mackay, E., Bahaj, A., Challenor, P.: Uncertainty in wave energy resource assessment. Part 2: Variability and predictability. Renew. Energy 35(8), 1809–1819 (2010)
- Agrawal, J., Deo, M.: On-line wave prediction. Mar. Struct. 15(1), 57–74 (2002)
- Sierra, J.P., White, A., Mösso, C., Mestres, M.: Assessment of the intraannual and inter-annual variability of the wave energy resource in the Bay of Biscay (France). Energy 141, 853–868 (2017)
- Penalba, M., Ulazia, A., Ibarra Berastegui, G., Ringwood, J., Sáenz, J.: Wave energy resource variation off the west coast of Ireland and its impact on realistic wave energy converters' power absorption. Appl. Energy 224, 205–219 (2018)
- Reguero, B., Losada, I., Méndez, F.: A recent increase in global wave power as a consequence of oceanic warming. Nat. Commun. 10(1), 1– 14 (2019)
- Nielsen, K., Pontes, T.: Generic and Site-Related Wave Energy Data (2010)
- Penalba, M., Ulazia, A., Saénz, J., Ringwood, J.V.: Impact of long-term resource variations on wave energy farms: the Icelandic case. Energy 192, (2020)

- Ulazia, A., Penalba, M., Ibarra Berastegui, G., Ringwood, J., Saénz, J.: Wave energy trends over the Bay of Biscay and the consequences for wave energy converters. Energy 141, 624–634 (2017)
- Ulazia, A., Penalba, M., Ibarra Berastegui, G., Ringwood, J., Sáenz, J.: Reduction of the capture width of wave energy converters due to longterm seasonal wave energy trends. Renew. Sustain. Energy Rev. 113, 109267 (2019)
- Cornett, A.M.: A global wave energy resource assessment. In: International Offshore and Polar Engineering Conference, pp. 1–9. ISOPE,Vancouver, CA (2008)
- Ringwood, J. & Brandle, G.: A new world map for wave power with a focus on variability. In: European Wave and Tidal Energy Conference, pp. 1–8. Nantes, France (2015)
- Coe, R.G., Ahn, S., Neary, V.S., Kobos, P.H., Bacelli, G.: Maybe less is more: considering capacity factor, saturation, variability, and filtering effects of wave energy devices. Appl. Energy 291, 116763 (2021)
- Lavidas, G., Blok, K.: Shifting wave energy perceptions: The case for wave energy converter (WEC) feasibility at milder resources. Renew. Energy 170, 1143–1155 (2021)
- Zubiate, L., Villate, J., Torre Enciso, Y., Soerensen, H., Holmes, B. & Panagiotopoulos, M. et al.: Methodology for site selection for wave energy projects. In: European Wave and Tidal Energy Conference, pp. 1089–1095. Glasgow, UK (2005)
- Wendt, F., Nielsen, K., Yu, Y.H., Bingham, H., Eskilsson, C., Kramer, M., et al.: Ocean energy systems wave energy modelling task: modelling, verification and validation of wave energy converters. J. Mar. Sci. Eng. 7(11), 379 (2019)
- Guo, B. & Ringwood, J.V.: Modelling of a vibro-impact power take-off mechanism for wave energy conversion. In: European Control Conference, pp. 1348–1353. IEEE, Saint Petersburg, Russia (2020)
- Guo, B., Ringwood, J.V.: Parametric study of a vibro-impact wave energy converter. IFAC-PapersOnLine 53(2), 12283–12288 (2020)
- Guo, B., Ringwood, J.V.: Non-linear modeling of a vibro-impact wave energy converter. IEEE Trans. Sust. Energy 12(1), 492–500 (2020)
- DualSPHysics.: DualSPHysics. https://dual.sphysics.org/ (2020). Accessed 10 Feb 2021
- Giorgi, G., Ringwood, J.V.: Nonlinear Froude-Krylov and viscous drag representations for wave energy converters in the computation/fidelity continuum. Ocean Eng. 141, 164–175 (2017)
- Ning, D.Z., Ke, S., Mayon, R., Zhang, C.: Numerical investigation on hydrodynamic performance of an OWC wave energy device in the stepped bottom. Front. Energy Res. 7, 1–11 (2019)
- Ning, D., Guo, B., Wang, R., Vyzikas, T., Greaves, D.: Geometrical investigation of a U-shaped oscillating water column wave energy device. Appl. Ocean Res. 97, 102105 (2020)
- Li, A., Liu, Y.: New analytical solutions to water wave diffraction by vertical truncated cylinders. Int. J. Nav. Archit. Ocean Eng. 11(2), 952–969 (2019)
- Li, A.j., Liu, Y., Li, H.j.: New analytical solutions to water wave radiation by vertical truncated cylinders through multi-term Galerkin method. Meccanica 54(3), 429–450 (2019)
- Zhao, H., Sun, Z., Hao, C., Shen, J.: Numerical modeling on hydrodynamic performance of a bottom-hinged flap wave energy converter. China Ocean Eng. 27(1), 73–86 (2013)
- Cummins, W.: The impulse response function and ship motions. Schiffstechnik 9, 101–109 (1962)
- Ogilvie, F.: Recent progress toward the understanding and prediction of ship motions. In: ONR Symposium on Naval Hydrodynamics, pp. 3–80. Bergen, Norway (1964)
- 111. Merigaud, A., Gilloteaux, J.C. & Ringwood, J.V.: A nonlinear extension for linear boundary element methods in wave energy device modelling. In: International Conference on Offshore Mechanics and Arctic Engineering, pp. 1–7. ASME, Rio de Janeiro, Brazil (2012).
- Penalba Retes, M., Mérigaud, A., Gilloteaux, J.C. & Ringwood, J.: Nonlinear Froude-Krylov force modelling for two heaving wave energy point absorbers. In: European Wave and Tidal Energy Conference, pp. 1–10. Nantes, France (2015)

- Giorgi, G., Ringwood, J.V.: Computationally efficient nonlinear Froude– Krylov force calculations for heaving axisymmetric wave energy point absorbers. J. Ocean Eng. Mar. Energy 3(1), 21–33 (2017)
- Giorgi, G., Ringwood, J.V.: Analytical representation of nonlinear Froude-Krylov forces for 3-DoF point absorbing wave energy devices. Ocean Eng. 164, 749–759 (2018)
- Morison, J., Johnson, J., Schaaf, S.: The force exerted by surface waves on piles. J. Pet. Technol. 2(05), 149–154 (1950)
- Gudmestad, O.T., Moe, G.: Hydrodynamic coefficients for calculation of hydrodynamic loads on offshore truss structures. Mar. Struct. 9(8), 745– 758 (1996)
- Giorgi, G. & Ringwood, J.: Consistency of viscous drag identification tests for wave energy applications. In: European Wave and Tidal Energy Conference, pp. 1–8. Cork, Ireland (2017)
- Guo, B., Patton, R., Jin, S., Gilbert, J., Parsons, D.: Nonlinear modeling and verification of a heaving point absorber for wave energy conversion. IEEE Trans. Sustain. Energy 9(1), 453–461 (2017)
- Guo, B., Patton, R.J.: Non-linear viscous and friction effects on a heaving point absorber dynamics and latching control performance. IFAC-PapersOnLine 50(1), 15657–15662 (2017)
- Jin, S., Patton, R.J., Guo, B.: Viscosity effect on a point absorber wave energy converter hydrodynamics validated by simulation and experiment. Renew. Energy 129, 500–512 (2018)
- Jin, S., Patton, R.J., Guo, B.: Enhancement of wave energy absorption efficiency via geometry and power take-off damping tuning. Energy 169, 819–832 (2019)
- Yu, Z., Falnes, J.: State-space modelling of a vertical cylinder in heave. Appl. Ocean Res. 17(5), 265–275 (1995)
- Taghipour, R., Perez, T., Moan, T.: Hybrid frequency-time domain models for dynamic response analysis of marine structure. Ocean Eng. 35(7), 685–705 (2008)
- Pérez, T., Fossen, T.: Time-vs. frequency-domain identification of parametric radiation force models for marine structure at zero speed. Model. Identif. Control 29(1), 1–19 (2008)
- Perez, T., Fossen, T.I.: A MATLAB toolbox for parametric identification of radiation-force models of ships and offshore structures. Model. Identif. Control 29(1), 1–15 (2009)
- Roessling, A., Ringwood, J.V.: Finite order approximations to radiation forces for wave energy applications. In: International Conference on Renewable Energies Offshore, pp. 359–366, Lison, Portugal (2014)
- Faedo, N., Peña Sanchez, Y., Ringwood, J.V.: Finite-order hydrodynamic model determination for wave energy applications using momentmatching. Ocean Eng. 163, 251–263 (2018)
- Abdelkhalik, O., Zou, S., Robinett, R., Bacelli, G., Wilson, D.: Estimation of excitation forces for wave energy converters control using pressure measurements. Int. J. Control. 90(8), 1793–1805 (2017)
- 129. Abdelrahman, M., Patton, R., Guo, B. & Lan, J.: Estimation of wave excitation force for wave energy converters. In: Conference on Control and Fault-Tolerant Systems, pp. 654–659. IEEE, Barcelona, Spain (2016)
- 130. Guo, B., Patton, R. & Jin, S.: Identification and validation of excitation force for a heaving point absorber wave energy convertor. In: European Wave and Tidal Energy Conference, pp. 1–9. Cork, Ireland (2017)
- 131. Giorgi, S., Davidson, J. & Ringwood, J.: Identification of nonlinear excitation force kernels using numerical wave tank experiments. In: European Wave and Tidal Energy Conference, pp. 1–10. Nantes, France (2015)
- Nguyen, H., Tona, P.: Wave excitation force estimation for wave energy converters of the point-absorber type. IEEE Trans. Control Syst. Technol. 26(6), 2173–2181 (2017)
- Lennart, L.: System Identification: Theory for the User. PTR Prentice Hall, Upper Saddle River, NJ (1999)
- Davidson, J., Giorgi, S., Ringwood, J.V.: Linear parametric hydrodynamic models for ocean wave energy converters identified from numerical wave tank experiments. Ocean Eng. 103, 31–39 (2015)
- 135. Giorgi, S., Davidson, J., Ringwood, J.V.: Identification of wave energy device models from numerical wave tank data'Part 2: data-based model determination. IEEE Trans. Sustain. Energy 7(3), 1020–1027 (2016)
- 136. Davidson, J., Giorgi, S., Ringwood, J.V.: Identification of wave energy device models from numerical wave tank data-Part 1: numerical wave

tank identification tests. IEEE Trans. Sustain. Energy 7(3), 1012–1019 (2016)

- Giorgi, S., Davidson, J., Jakobsen, M., Kramer, M., Ringwood, J.V.: Identification of dynamic models for a wave energy converter from experimental data. Ocean Eng. 183, 426–436 (2019)
- Babarit, A.: Ocean Wave Energy Conversion: Resource, Technologies and Performance. Elsevier, Amsterdam (2017)
- Rafiee, A. & Fiévez, J.: Numerical prediction of extreme loads on the CETO wave energy converter. In: European Wave and Tidal Energy Conference, pp. 1–10. Nantes, France (2015)
- Ransley, E.J., Greaves, D., Raby, A., Simmonds, D., Hann, M.: Survivability of wave energy converters using CFD. Renew. Energy 109, 235–247 (2017)
- Garcia Rosa, P.B., Bacelli, G., Ringwood, J.V.: Control-informed optimal array layout for wave farms. IEEE Trans. Sustain. Energy 6(2), 575–582 (2015)
- 142. Penalba, M., Touzón, I., Lopez Mendia, J., Nava, V.: A numerical study on the hydrodynamic impact of device slenderness and array size in wave energy farms in realistic wave climates. Ocean Eng. 142, 224–232 (2017)
- Fang, H.W., Feng, Y.Z., Li, G.P.: Optimization of wave energy converter arrays by an improved differential evolution algorithm. Energies 11(3522), 1–19 (2018)
- Lyu, J., Abdelkhalik, O., Gauchia, L.: Optimization of dimensions and layout of an array of wave energy converters. Ocean Eng. 192, 106543 (2019)
- Child, B.: On the configuration of arrays of floating wave energy converters. Dissertation, University of Edinburgh (2011)
- Penalba, M., Sell, N.P., Hillis, A.J., Ringwood, J.V.: Validating a wave-towire model for a wave energy converte–Part I: the hydraulic transmission system. Energies 10(7), 977 (2017)
- Penalba, M., Cortajarena, J.A., Ringwood, J.V.: Validating a wave-to-wire model for a wave energy converter–Part II: the electrical system. Energies 10(7), 1002 (2017)
- Henderson, R.: Design, simulation, and testing of a novel hydraulic power take-off system for the Pelamis wave energy converter. Renew. Energy 31(2), 271–283 (2006)
- Kurniawan, A., Pedersen, E., Moan, T.: Bond graph modelling of a wave energy conversion system with hydraulic power take-off. Renew. Energy 38(1), 234–244 (2012)
- Vantorre, M., Banasiak, R., Verhoeven, R.: Modelling of hydraulic performance and wave energy extraction by a point absorber in heave. Appl. Ocean Res. 26(1-2), 61–72 (2004)
- Kim, S.J., Koo, W., Shin, M.J.: Numerical and experimental study on a hemispheric point-absorber-type wave energy converter with a hydraulic power take-off system. Renew. Energy 135, 1260–1269 (2019)
- Cargo, C., Hillis, A., Plummer, A.: Strategies for active tuning of wave energy converter hydraulic power take-off mechanisms. Renew. Energy 94, 32–47 (2016)
- Falcão, A., Gato, L., Nunes, E.: A novel radial self-rectifying air turbine for use in wave energy converters. Renew. Energy 50, 289–298 (2013)
- Kofoed, J.P., Frigaard, P., Friis Madsen, E., Sørensen, H.C.: Prototype testing of the wave energy converter wave dragon. Renew. Energy 31(2), 181–189 (2006)
- Li, X., Chen, C., Li, Q., Xu, L., Liang, C., Ngo, K., et al.: A compact mechanical power take-off for wave energy converters: Design, analysis, and test verification. Appl. Energy 278, 115459 (2020)
- Liang, C., Ai, J., Zuo, L.: Design, fabrication, simulation and testing of an ocean wave energy converter with mechanical motion rectifier. Ocean Eng. 136, 190–200 (2017)
- Mueller, M., Baker, N.: Direct drive electrical power take-off for offshore marine energy converters. P. I. Mech. Eng. A-J. Pow. 219(3), 223–234 (2005)
- Rhinefrank, K., Schacher, A., Prudell, J., Brekken, T.K., Stillinger, C., Yen, J.Z., et al.: Comparison of direct-drive power takeoff systems for ocean wave energy applications. IEEE J. Ocean Eng. 37(1), 35–44 (2011)

- Faiz, J., Nematsaberi, A.: Linear electrical generator topologies for directdrive marine wave energy conversion-an overview. IET Renew. Power Gener. 11(9), 1163–1176 (2017)
- Têtu, A.: Power take-off systems for WECS. In: Pecher, A., Kofoed, J. (eds.) Handbook of Ocean Wave Energy, pp. 203–220. Springer, Cham, (2017)
- Falcão, A., Henriques, J.: Effect of non-ideal power take-off efficiency on performance of single-and two-body reactively controlled wave energy converters. J. Ocean Eng. Mar. Energy 1(3), 273–286 (2015)
- 162. de la Villa Jaén, A., Santana, A.G., et al.: Considering linear generator copper losses on model predictive control for a point absorber wave energy converter. Energy Convers. Manage. 78, 173–183 (2014)
- Mueller, M.: Electrical generators for direct drive wave energy converters. IEE Proc.-Gener., Transm. Distrib. 149(4), 446–456 (2002)
- Bacelli, G., Genest, R., Ringwood, J.V.: Nonlinear control of flap-type wave energy converter with a non-ideal power take-off system. Annu. Rev. Control 40, 116–126 (2015)
- Penalba, M., Ringwood, J.V.: A high-fidelity wave-to-wire model for wave energy converters. Renew. Energy 134, 367–378 (2019)
- Penalba, M., Ringwood, J.V.: Systematic complexity reduction of waveto-wire models for wave energy system design. Ocean Eng. 217, 107651 (2020)
- Faedo, N., Piuma, F.J.D., Giorgi, G., Ringwood, J.V.: Nonlinear model reduction for wave energy systems: a moment-matching-based approach. Nonlinear Dyn. 102, 1215–1237 (2020)
- Budar, K., Falnes, J.: A resonant point absorber of ocean-wave power. Nature 256(5517), 478–479 (1975)
- Fusco, F., Ringwood, J.V.: A simple and effective real-time controller for wave energy converters. IEEE Trans. Sustain. Energy 4(1), 21–30 (2012)
- Davidson, J., Genest, R., Ringwood, J.V.: Adaptive control of a wave energy converter. IEEE Trans. Sustain. Energy 9(4), 1588–1595 (2018)
- Babarit, A., Guglielmi, M., Clément, A.H.: Declutching control of a wave energy converter. Ocean Eng. 36(12-13), 1015–1024 (2009)
- 172. Ringwood, J.V., Mérigaud, A., Faedo, N., Fusco, F.: An analytical and numerical sensitivity and robustness analysis of wave energy control systems. IEEE Trans. Control Syst. Technol. 28(4), 1337–1348 (2019)
- Hals, J., Falnes, J., Moan, T.: Constrained optimal control of a heaving buoy wave-energy converter. J. Offshore Mech. Arch. 133(1), (2011)
- Brekken, T.K.: On model predictive control for a point absorber wave energy converter. In: IEEE PES Trondheim PowerTech, pp. 1–8. IEEE, Trondheim, Norway (2011)
- 175. Amann, K.U., Magaña, M.E., Sawodny, O.: Model predictive control of a nonlinear 2-body point absorber wave energy converter with estimated state feedback. IEEE Trans. Sustain. Energy 6(2), 336–345 (2014)
- O'Sullivan, A.C., Lightbody, G.: Co-design of a wave energy converter using constrained predictive control. Renew. Energy 102, 142–156 (2017)
- Li, G., Belmont, M.R.: Model predictive control of sea wave energy converters–Part II: the case of an array of devices. Renew. Energy 68, 540–549 (2014)
- Bacelli, G., Ringwood, J.V.: A geometric tool for the analysis of position and force constraints in wave energy converters. Ocean Eng. 65, 10–18 (2013)
- Fusco, F. & Ringwood, J.V.: Robust control of wave energy converters. In: IEEE Conference on Control Applications, pp. 292–297. IEEE, Juan Les Antibes, France (2014)
- Na, J., Li, G., Wang, B., Herrmann, G., Zhan, S.: Robust optimal control of wave energy converters based on adaptive dynamic programming. IEEE Trans. Sustain. Energy 10(2), 961–970 (2018)
- García Violini, D. & Ringwood, J.V.: Robust control of wave energy converters using spectral and pseudospectral methods: a case study. In: American Control Conference, pp. 4779–4784. IEEE, Philadelphia, PA, USA (2019)
- Lao, Y. & Scruggs, J.T.: Robust control of wave energy converters using unstructured uncertainty. In: American Control Conference, pp. 4237– 4244. IEEE, Denver, CO, USA (2020)
- Faedo, N., García Violini, D., Scarciotti, G., Astolfi, A. & Ringwood, J.V.: Robust moment-based energy-maximising optimal control of wave

energy converters. In: IEEE Conference on Decision and Control, pp. 4286–4291. IEEE, Nice, France (2019)

- 184. Garrido, A.J., Garrido, I., Alberdi, M., Amundarain, M., Barambones, O. & Romero, J.A.: Robust control of oscillating water column (OWC) devices: power generation improvement. In: OCEANS-San Diego, pp. 1–4. IEEE, San Diego, USA (2013)
- 185. Wahyudie, A., Jama, M., Saeed, O., Noura, H., Assi, A., Harib, K.: Robust and low computational cost controller for improving captured power in heaving wave energy converters. Renew. Energy 82, 114–124 (2015)
- Zhan, S., He, W., Li, G.: Robust feedback model predictive control of sea wave energy converters. IFAC-PapersOnLine 50, 141–146, (2017)
- 187. Fusco, F. & Ringwood, J.: A model for the sensitivity of non-causal control of wave energy converters to wave excitation force prediction errors. In: European Wave and Tidal Energy Conference, pp. 1–10. Southampton, UK (2011)
- 188. Parrinello, L., Dafnakis, P., Pasta, E., Bracco, G., Naseradinmousavi, P., Mattiazzo, G., et al.: An adaptive and energy-maximizing control optimization of wave energy converters using an extremum-seeking approach. Phys. Fluids 32(11), 113307 (2020)
- Valério, D., Mendes, M.J., Beirão, P., da Costa, J.S.: Identification and control of the AWS using neural network models. Appl. Ocean Res. 30(3), 178–188 (2008)
- Anderlini, E., Husain, S., Parker, G.G., Abusara, M., Thomas, G.: Towards real-time reinforcement learning control of a wave energy converter. J. Mar. Sci. Eng. 8(11), 845 (2020)
- 191. Thomas, S., Giassi, M., Eriksson, M., Göteman, M., Isberg, J., Ransley, E., et al.: A model free control based on machine learning for energy converters in an array. Big Data Cognit. Comput. 2(4), 36 (2018)
- 192. Anderlini, E., Forehand, D.I., Bannon, E., Abusara, M.: Control of a realistic wave energy converter model using least-squares policy iteration. IEEE Trans. Sustain. Energy 8(4), 1618–1628 (2017)
- Bacelli, G., Balitsky, P., Ringwood, J.V.: Coordinated control of arrays of wave energy devices–Benefits over independent control. IEEE Trans. Sustain. Energy 4(4), 1091–1099 (2013)
- Li, G. & Belmont, M.: Model predictive control of an array of wave energy converters. In: European Wave and Tidal Energy Conference, pp. 1–10. Aalborg, Denmark (2013)
- 195. Forehand, D., Kiprakis, A., Nambiar, A., Wallace, R.: A fully coupled wave-to-wire model of an array of wave energy converters. IEEE Trans. Sustain. Energy 7(1), 118–128 (2015)
- 196. Faedo, N., Scarciotti, G., Astolfi, A. & Ringwood, J.V.: Moment-based constrained optimal control of an array of wave energy converters. In: American Control Conference, pp. 4797–4802. IEEE, Philadelphia, PA, USA (2019)
- Zhong, Q., Yeung, R.W.: Model-predictive control strategy for an array of wave-energy converters. J. Mar. Sci. Appl. 18(1), 26–37 (2019)
- Fitzgerald, J. & Bolund, B.: Technology readiness for wave energy projects; ESB and Vattenfall classification system. In: International Conference on Ocean Energy, pp. 17–19. Dublin, Ireland (2012)
- Ruehl, K. & Bull, D.: Wave energy development roadmap: design to commercialization. In: Oceans, pp. 1–10. IEEE, Hampton Roads, VA, USA (2012)
- Holmes, B., Nielsen, K.: Guidelines for the Development & Testing of Wave Energy Systems (2010)
- Weber, J.: Wec technology readiness and performance matrix–finding the best research technology development trajectory. In: International Conference on Ocean Energy, pp. 1–10 (2012)
- Weber, J., Costello, R. & Ringwood, J.: WEC technology performance levels (TPLs)-metric for successful development of economic WEC technology. In: European Wave and Tidal Energy Conference, pp. 1–9. Aalborg, Denmark (2013)
- Weber, J., Roberts, J.D., Babarit, A., Costello, R., Bull, D.L., Neilson, K., et al.: Guidance on the Technology Performance Level (TPL) Assessment Methodology (2016)
- Gunawan, B.: Wave Energy Prize: Testing and Data Analysis Overview. U.S. Department of Energy, Washington, DC (2018)
- Roberts, J.D., Bull, D.L., Malins, R.J., Weber, J., Dykes, K., Babarit, A., et al.: Technology Performance Level (TPL) Assessment (2017)

- Bull, D.L., Costello, R.P., Babarit, A., Kim, N., Kennedy, B., et al.: Scoring the Technology Performance Level (TPL) Assessment (2017)
- ARENA: Technology Readiness Levels for Renewable Energy Sectors (2014)
- ARENA.: Commercial readiness index for renewable energy sectors. (Australian Renewable Energy Agency.) https://arena.gov.au/assets/ 2014/02/Commercial-Readiness-Index.pdf (2014). Accessed 10 Feb 2021
- Murphy, L.M., Edwards, P.L.: Bridging the Valley of Death: Transitioning from Public to Private Sector Financing. National Renewable Energy Laboratory, Golden, CO (2003)
- Clément, A., McCullen, P., Falcão, A., Fiorentino, A., Gardner, F., Hammarlund, K., et al.: Wave energy in Europe: current status and perspectives. Renew. Sustain. Energy Rev. 6(5), 405–431 (2002)
- 211. Salter, S.: Wave power. Nature 249(5459), 720-724 (1974)
- 212. Evans, D.V.: A theory for wave-power absorption by oscillating bodies. J. Fluid Mech. 77(1), 1–25 (1976)
  212. Public Film and Control of the second second
- Budal, K., Falnes, J.: Optimum operation of improved wave-power converter. Mar. Sci. Commun. 3(2), 133–159 (1977)
- Salter, S., Jeffrey, D. & Taylor, J.: The architecture of nodding duck wave power generators. Naval Architect 1, 21–24 (1976)
- Thomas, G., Evans, D.: Arrays of three-dimensional wave-energy absorbers. J. Fluid Mech. 108, 67–88 (1981)
- Leishman, J. & Scobie, G.: The Development of Wave Power: A Techno-Economic Study, Dept. of the Industry, National Engineering Laboratory (1976)
- 217. Naito, S., Nakamura, S.: Wave energy absorption in irregular waves by feedforward control system. In: Evan, D.V., de Falcso, A.F.O. (eds.) Hydrodynamics of Ocean Wave-Energy Utilization, pp. 269–280. Springer, Berlin (1986)
- Mehlum, E.: Tapchan. In: Hydrodynamics of Ocean Wave-Energy Utilization, pp. 51–55. Springer, Berlin (1986)
- Malmo, O., Reitan, A.: Development of the Kvaerner multiresonant OWC. In: Hydrodynamics of Ocean Wave-Energy Utilization, pp. 57–67. Springer, Berlin (1986)
- 220. Falcão, A., Sarmento, A., Gato, L., Brito Melo, A.: The Pico OWC wave power plant: its lifetime from conception to closure 1986–2018. Appl. Ocean Res. 98, 102104 (2020)
- Jin, S., Greaves, D.: Wave energy in the UK: status review and future perspectives. Renew. Sustain. Energy Rev. 143, 110932 (2021)
- Masuda, Y., McCormick, M.E.: Experiences in pneumatic wave energy conversion in Japan. In: Utilization of ocean waves–wave to energy conversion, pp. 1–33. ASCE, Reston, VA (1986)
- Fredriksen, A.: Tapered channel wave power plants. In: Energy for Rural and Island Communities, pp. 179–182. Elsevier, Amsterdam (1986)
- Heath, T.: The development and installation of the LIMPET Wave Energy Converter. In: World Renewable Energy Congress, pp. 1619– 1622. Elsevier, Amsterdam (2000)
- 225. Washio, Y., Osawa, H., Nagata, Y., Fujii, F., Furuyama, H. & Fujita, T.: The offshore floating type wave power device "Mighty Whale": open sea tests. In: International Offshore and Polar Engineering Conference, pp. 373–380. ISOPE, Seattle, Washington, USA (2000)
- 226. Wu, B., Chen, T., Jiang, J., Li, G., Zhang, Y., Ye, Y.: Economic assessment of wave power boat based on the performance of "Mighty Whale" and BBDB. Renew. Sustain. Energy Rev. 81, 946–953 (2018)
- 227. Boake, C.B., Whittaker, T.J., Folley, M. & Ellen, H.: Overview and initial operational experience of the LIMPET wave energy plant. In: International Offshore and Polar Engineering Conference, pp. 586–594. ISOPE, KitaKyushu, Japan (2002)
- Prado, M. & Polinder, H.: Direct drive in wave energy conversion–AWS full scale prototype case study. In: IEEE Power and Energy Society General Meeting, pp. 1–7. IEEE, Detroit, MI, USA (2011)
- Prado, M., Polinder, H.: Case study of the Archimedes Wave Swing (AWS) direct drive wave energy pilot plant. In: Electrical Drives for Direct Drive Renewable Energy Systems, pp. 195–218. Elsevier, Amsterdam (2013)

- 230. Yemm, R., Pizer, D., Retzler, C., Henderson, R.: Pelamis: experience from concept to connection. Philos. Trans. Royal. Soc. A 370(1959), 365-380 (2012)
- 231. Leijon, M., Boström, C., Danielsson, O., Gustafsson, S., Haikonen, K., Langhamer, O., et al.: Wave energy from the North Sea: experiences from the Lysekil research site. Surv. Geophys. 29(3), 221-240 (2008)
- 232. Lejerskog, E., Boström, C., Hai, L., Waters, R., Leijon, M.: Experimental results on power absorption from a wave energy converter at the Lysekil wave energy research site. Renew. Energy 77, 9-14 (2015)
- 233. Kasanen, E.: AW-Energy-Positive experiences of the Waveroller in Portugal and France. In: WavEC Annual Seminar and International B2B Meetings, pp. 15-17. Lisbon, Portugal (2015)
- 234. Yu, Y.H. & Li, Y.: A RANS simulation of the heave response of a twobody floating-point wave absorber. In: International Offshore and Polar Engineering Conference, pp. 1–7. Maui, Hawaii, USA (2011)
- 235. Torre Enciso, Y., Ortubia, I., De Aguileta, L.L. & Marqués, J.: Mutriku wave power plant: from the thinking out to the reality. In: European Wave and Tidal Energy Conference, Uppsala, Sweden, vol. 710, pp. 319-329. Uppsala, Sweden (2009)
- 236. Ibarra Berastegi, G., Sáenz, J., Ulazia, A., Serras, P., Esnaola, G., Garcia Soto, C.: Electricity production, capacity factor, and plant efficiency index at the Mutriku wave farm (2014-2016). Ocean Eng. 147, 20-29 (2018)
- 237. Boyle, L., Doherty, K., van't Hoff, J. & Skelton, J.: The value of full scale prototype data-testing oyster 800 at EMEC, Orkney. In: European Wave and Tidal Energy Conference, pp. 6-11. Nantes, France (2015)
- 238. Joslin, J., Cotter, E., Murphy, P., Gibbs, P., Cavagnaro, R. & Crisp, C. et al.: The wave-powered adaptable monitoring package: hardware design, installation, and deployment. In: European Wave and Tidal Energy Conference, pp. 1-6. Napoli, Italy (2019)
- 239. Cross, P.: Recent Developments at the U.S. Navy Wave Energy Test Site. In: European Wave and Tidal Energy Conference, pp. 1-9. Napoli, Italy (2019)
- 240. ARENA .: Oceanlinx 1MW commercial wave energy demonstrator. (Australian Renewable Energy Agency.) https://arena.gov.au/ projects/oceanlinx-1mw-commercial-wave-energy-demonstrator/ (2021). Accessed 10 Feb 2021.
- 241. Sheng, S., Wang, K., Lin, H., Zhang, Y., You, Y., Wang, Z., et al.: Model research and open sea tests of 100 kW wave energy convertor Sharp Eagle Wanshan. Renew. Energy 113, 587-595 (2017)
- 242. Greaves, D., Iglesias, G.: Wave and Tidal Energy. John Wiley & Sons, Hoboken, NJ (2018)
- 243. Chatzigiannakou, M.A., Dolguntseva, I., Leijon, M.: Offshore deployments of wave energy converters by seabased industry AB. J. Mar. Sci. Eng. 5(2), 15 (2017)
- 244. Yu, Y.H., Li, Y.: Reynolds-averaged Navier-Stokes simulation of the heave performance of a two-body floating-point absorber wave energy system. Comput. Fluids 73, 104-114 (2013)
- 245. De Andres, A., Medina Lopez, E., Crooks, D., Roberts, O., Jeffrey, H.: On the reversed LCOE calculation: design constraints for wave energy commercialization. Int. J. Mar. Energy 18, 88-108 (2017)
- 246. Tran, T.T., Smith, A.D.: Incorporating performance-based global sensitivity and uncertainty analysis into LCOE calculations for emerging renewable energy technologies. Appl. Energy 216, 157-171 (2018)
- 247. OES: Annual Report: An Overview of Ocean Energy Activities in 2019 (2019)
- 248. IEA : Blue Economy and its Promising Markets for Ocean Energy (2020)
- 249. Green, R., Copping, A., Cavagnaro, R., Rose, D., Overhus, D., Jenne, D.: Enabling power at sea: opportunities for expanded ocean observations through marine renewable energy integration. In: OCEANS 2019 Seattle, pp. 1-7. IEEE, Piscataway, NJ (2019)
- 250. Xu, C., Huang, Z.: A dual-functional wave-power plant for wave-energy extraction and shore protection: a wave-flume study. Appl. Energy 229, 963-976 (2018)

- 251. Zheng, S., Antonini, A., Zhang, Y., Greaves, D., Miles, J., Iglesias, G.: Wave power extraction from multiple oscillating water columns along a straight coast. J. Fluid Mech. 878, 445-480 (2019)
- 252. Zheng, S., Zhang, Y., Iglesias, G.: Coast/breakwater-integrated OWC: a theoretical model. Mar. Struct. 66, 121-135 (2019)
- 253. Corsini, A., Tortora, E., Cima, E.: Preliminary assessment of wave energy use in an off-grid minor island desalination plant. Energy Procedia 82, 789-796 (2015)
- 254. Leijon, J., Boström, C.: Freshwater production from the motion of ocean waves-a review. Desalination 435, 161-171 (2018)
- 255. Oliveira Pinto, S., Rosa Santos, P., Taveira Pinto, F.: Electricity supply to offshore oil and gas platforms from renewable ocean wave energy: Overview and case study analysis. Energy Convers. Manage. 186, 556-569 (2019)
- 256. Tay, Z.Y.: Energy extraction from an articulated plate anti-motion device of a very large floating structure under irregular waves. Renew. Energy 130, 206-222 (2019)
- 257. LiVecchi, A., Copping, A., Jenne, D., Gorton, A., Preus, R., Gill, G., et al.: Powering the Blue Economy; Exploring Opportunities for Marine Renewable Energy in Maritime Markets. U.S. Department of Energy, Washington, DC (2019)
- 258. GWEC: Global Offshore Wind Report 2020 (2020)
- 259. Díaz, H., Soares, C.G.: Review of the current status, technology and future trends of offshore wind farms. Ocean Eng. 209, 107381 (2020)
- 260. OES: Annual Report: An Overview of Ocean Energy Activities in 2017 (2017)
- 261. Supergen: Wave energy road map. (Supergen Offshore Renewhttps://epsrc.ukri.org/files/funding/calls/2020/ able Energy.) wave-energy-road-map/ (2020). Accessed 10 Feb 2021
- 262. Nielsen, K., Krogh, J., Brodersen, H.J., Steenstrup, P., Pilgaard, H., Marquis, L., et al.: Partnership for wave power-roadmaps. DCE Technical Report No. 186 (2015)
- 263. SEAL: Ocean Energy Roadmap Ireland. (Sustainable Energy Authority of Ireland.) https://www.ocean-energy-systems.org/publications/ national-roadmaps/ (2015). Accessed 10 Feb 2021
- 264. Dalton, G., Ó Gallachóir, B.P.: Building a wave energy policy focusing on innovation, manufacturing and deployment. Renew. Sustain. Energy Rev. 14(8), 2339-2358 (2010)
- 265. Vining, J.G., Muetze, A.: Economic factors and incentives for ocean wave energy conversion. IEEE Trans. Ind. Appl. 45(2), 547-554 (2009)
- 266. Lu, Y., Khan, Z.A., Alvarez Alvarado, M.S., Zhang, Y., Huang, Z., Imran, M.: A critical review of sustainable energy policies for the promotion of renewable energy sources. Sustainability 12(12), 5078 (2020)
- 267. Dalton, G.J., Alcorn, R., Lewis, T.: Case study feasibility analysis of the Pelamis wave energy convertor in Ireland, Portugal and North America. Renew. Energy 35(2), 443-455 (2010)
- 268. OES: Annual Report: An Overview of Ocean Energy Activities in 2018 (2018)
- 269. Ma, C.L.: Ocean energy management policy in China. Procedia Comput. Sci. 154, 493-499 (2019)

How to cite this article: Guo, B., Ringwood, J.V. A review of wave energy technology from a research and commercial perspective. IET Renew. Power Gener. 2021;15:3065-3090.

https://doi.org/10.1049/rpg2.12302